

THE CASE FOR A 'BUCKET APPROACH' TO FISCAL MULTIPLIERS AND MORE

WRITTEN BY: DOMINIC CADDICK, CHAITANYA KUMAR



# CONTENTS

EXECUTIVE SUMMARY				
1. INTRODUCTION	4			
<ul> <li>2. WHAT ARE FISCAL MULTIPLIERS?</li> <li>2.1 HOW CAN WE MEASURE MULTIPLIERS?</li> <li>2.2 WHAT AFFECTS THE SIZE OF MULTIPLIERS?</li> <li>2.3 HOW DOES ECONOMIC CONTEXT AFFECT MULTIPLIERS?</li> <li>2.4 HOW LONG DO MULTIPLIERS LAST?</li> </ul>	<b>6</b> 7 10 11			
<ul> <li>3. HOW DO GOVERNMENTS FORECAST THE IMPACT OF FISCAL POLICY?</li> <li>3.1 CASE STUDIES OF FISCAL FORECASTERS</li> <li>3.2 THE IMPOSSIBLE NEUTRALITY OF FISCAL FORECASTERS</li> <li>3.3 THE INTERACTION WITH FISCAL RULES</li> </ul>	<b>14</b> 14 16 18			
<ul> <li>4. THE CASE FOR A BUCKET APPROACH TO DECIDE FISCAL MULTIPLIERS</li> <li>4.1 THE IMF'S BUCKET APPROACH BY COUNTRY</li> <li>4.2 NEF'S BUCKET APPROACH BY POLICY</li> <li>4.3 APPLYING THE BUCKET APPROACH: THE 2024 AUTUMN STATEMENT</li> <li>4.4 THE BENEFITS OF THE BUCKET APPROACH</li> </ul>	<b>20</b> 20 20 24 26			
<ul> <li>5. FURTHER RECOMMENDATIONS TO IMPROVE FORECASTING THE EFFECTS OF POLICY</li> <li>5.1 EMBRACING UNCERTAINTY WITH RANGES</li> <li>5.2 SEEING THE FULL BENEFIT WITH LONGER HORIZONS</li> <li>5.3 A BUCKET FOR CLIMATE, INEQUALITY, AND MORE</li> </ul>	<b>28</b> 28 28 29			
6. CONCLUSION	30			

### EXECUTIVE SUMMARY

iscal forecasting in the UK currently relies on rigid fiscal multiplier assumptions that constrain effective government policy. Fiscal multipliers, which measure the impact of government spending on gross domestic product (GDP), are central to economic forecasting but are applied too narrowly, limiting the perceived benefits of public investment. This is particularly so in areas essential for long-term growth like public services, green infrastructure, and social equity. This static, onesize-fits-all approach restricts the scope for targeted government intervention, reinforcing a cycle of low investment and low growth while undervaluing policies that address critical structural issues such as climate change, inequality, and economic resilience. The result is a forecasting model that inherently favours fiscal restraint, discouraging investment that could foster a more sustainable and equitable economy.

In this report, we present a brief assessment of the Office for Budget Responsibility's (OBR) analysis of the 2024 Autumn Statement, where significant public investment was projected to yield only a 0.15% GDP growth by 2029-30. Despite the government's planned 2.2% GDP increase in spending, the OBR's assumptions - based on narrow multiplier applications and limited longterm impact - predicted minimal economic gains, emphasising crowding-out effects over potential productivity and demand-side benefits. This conservative approach devalues public spending's potential, limiting the perceived returns on investment even in sectors with high multipliers, such as green technology or social infrastructure. By embedding low multiplier effects in its analysis, the current model prioritises short-term fiscal targets over the longer-term economic and social gains that targeted government spending could achieve.

To address these limitations, the report proposes a new'bucket approach' to fiscal multipliers, providing a more flexible and context-sensitive model. This method, inspired by International Monetary Fund (IMF) practices, categorises policies based on specific characteristics rather than applying blanket multipliers. Policies are assessed on factors known to influence multiplier effects, including the likelihood of stimulating consumption among those with high marginal propensities to spend, creating demand in industries with significant domestic supply chains, and generating immediate economic activity through direct government expenditure rather than tax cuts. This approach also accounts for policies that could encourage private investment, expand or improve the productive capacity of the economy, and reduce barriers to productivity growth.

Based on these characteristics, policies are grouped into multiplier 'buckets' that correspond to estimated ranges, with adjustments for economic context. For example, high-scoring policies are allocated higher multiplier ranges to reflect the broader, more enduring impacts they are expected to generate. By contrast, policies with lower scores fall into lower multiplier ranges, capturing their limited potential to stimulate the economy. This scoring system captures the varied economic impacts of different types of government spending and allows for a dynamic approach to multipliers, where adjustments can be made based on how current economic conditions are judged, such as the size of the output gap or changes in monetary policy stance. In periods of economic slack, for instance, multiplier effects can be scaled up to reflect the greater potential for government spending to drive growth.

This approach increases transparency by clarifying the basis for each policy's multiplier and ultimately enables more informed public debate. By allowing adjustments to multiplier ranges based on policy characteristics and economic context, we minimise the risk of misrepresenting impacts, reducing reliance on outdated averages that may not capture present realities.

More up-to-date multiplier assumptions will invariably create greater fiscal space, enabling more extensive public investment in vital areas, from public services to green transition initiatives, even within restrictive fiscal rules. Furthermore, this flexibility could encourage replacing fixed fiscal rules with a more holistic system of fiscal referees<sup>1</sup> whose judgments could usefully discern if the economic effects needed for a policy package to avoid debt sustainability risks were realistic. Moving beyond rigid multipliers, the bucket approach helps align fiscal planning with broader social and environmental goals and acts as a tool to better manage uncertainty, something that a mission-led government needs to consider more seriously.

### **1. INTRODUCTION**

n the next decade, it is critical that governments use economic policy to tackle the many social and environmental crises of our time. Yet a reliance on fiscal rules that limit public borrowing decisions and try to cap government debt is likely holding us back from making progress. Furthermore, the fiscal forecasters who decide if these rules are met operate on outdated assumptions about how government spending impacts the economy – making it impossible to even forecast ourselves out of low growth.

In fact, at her first budget as Chancellor, Rachel Reeves may have already had a taste of this. Despite announcing a package of taxes and £372bn extra spending on public services and investment until 2030 - an increase in spending of 2.2% of gross domestic product (GDP) by 2029–30,<sup>2</sup> the Office for Budget Responsibility (OBR; the UK's fiscal forecaster) only forecast this to increase GDP in 2029-30 by 0.15% compared to if no spending had taken place at all. The OBR assumes the effects of tax measures worth 1.1% of GDP and investment crowding out drag GDP down by 0.26%. This meant that the total effect of government policy by 2029–30 for the economy was to be 0.11% smaller than if no policies were implemented at all.

Such a result isn't necessarily surprising, as we wrote at NEF before the budget.<sup>3</sup> The outdated assumptions the OBR uses necessitated any large increase in spending to only be seen as a temporary sugar rush.<sup>4</sup> Yet, it is worth reflecting that when previous governments announced austerity measures, which Labour's plans do not even fully reverse,<sup>5</sup> the OBR's forecasts were much rosier. Therefore, one must hope the OBR is wrong again and, instead, Labour's investment in our public services will make a material difference to our lives and the economy, making the case for even more spending to finally make austerity a thing of the past.

Since the budget, the OBR's forecasts have caused a stir, with many arguing that it is likely underestimating the growth impact of investment.<sup>6,7</sup> In fact, the problem might be even wider-ranging than what is traditionally considered investment. For example, NEF research has shown how scrapping the two-child limit and benefit cap on welfare payments could lower the cost of child poverty by £3bn a year.<sup>8</sup>Yet if the OBR's standard assumptions are applied then the economic effect of providing support to those in the most acute poverty would be assumed to fade after five years. Saying this, it is important to reflect that economics is not a science, and the OBR's judgment of the economic literature could be entirely different. Yet the OBR does not seem to robustly justify itself and key assumptions are left unexplained and unjustified. Therefore, this report tackles this question: How can forecasters be more transparent in their assumptions and best reflect current evidence?

At the heart of this debate is the fiscal multiplier. That is, how much of an effect a change in government spending (fiscal policy) can have on GDP. The larger the fiscal multiplier, the more government policy affects GDP. This is important because when fiscal forecasters make assumptions about the size and longevity of the fiscal multiplier, they bake in certain outcomes. For example, many governments have fiscal rules that limit their debtto-GDP ratios. Yet when governments borrow to spend they affect both the level of debt by how much is borrowed and the level of GDP through the multiplier on spending. We calculate that if the OBR used higher multiplier assumptions it would have allowed the chancellor to spend up to £15.6bn more per year without the debt-to-GDP ratio rising.

We must challenge the economic orthodoxies headon as the assumptions forecasters make may be a fruitful area where small changes can make a big difference. This is especially true at a moment when the economy needs to make unprecedented levels of investment to protect the planet from climate change.9 At this moment, it feels as if the past will be a particularly bad guide for the future. Yet we don't have to go into this moment completely blind. By learning from the most effective economic policies of the past we can make good guesses of how policy might affect the economy while embracing the fundamental uncertainty at the same time. If the possibility of government spending and investment making a significant difference to the economy is devalued, then it will be harder

for governments to act on ambitions even with democratic mandates. To escape this doom-loop<sup>10</sup> we must find a framework that is flexible enough to see the upsides of government spending and honest enough to see the uncertainty.

To explore whether a better forecasting framework is possible, this report is set out as follows. Chapter 2 looks at the theoretical foundations of multipliers and explores contemporary empirical evidence. Chapter 3 looks at the use of multipliers in practice by different government forecasters. Chapter 4 proposes a bucket approach to multipliers inspired by a method practised by the International Monetary Fund (IMF). Chapter 5 sets out additional ways to improve forecasting frameworks, and Chapter 6 concludes.

### 2. WHAT ARE FISCAL MULTIPLIERS?

deally, governments make spending decisions that improve people's lifelong wellbeing. When governments make spending cuts, they can sacrifice our wellbeing by dismantling safety-nets, reducing the quality of the services we rely on, and jeopardising long-term prosperity by failing to provide the infrastructure and skills needed for the future. Therefore, it begs the question: What is the extent that government spending decisions affect our lives?

There are myriad ways we could measure the impact of government spending, but a good place to start is the work of renowned economists Richard Kahn and John Maynard Keynes in the 1930s.

Kahn sought to quantify the tangible benefits of government investment, building on the longstanding economic observation that investment decisions often directly lead to the creation of new job opportunities, which helps move people onto higher incomes or transition from unemployment to employment.<sup>11</sup> This increase in income can generate secondary indirect effects, as this new spending travels down supply chains, for example, affecting not just the bricklayers but the brickmakers too. Together, these workers spend their newly acquired wages, thereby generating additional income for others who can then spend their own increased earnings inducing effects across the wider economy, potentially creating jobs in areas entirely unrelated to the initial investment. Equally, if investment just causes people to change jobs or people do not spend their new wages then there may be little direct, indirect, or induced effects to be seen. Kahn set out the case that when the economy has sufficient spare capacity, for people to consume more or move into employment, then the number of jobs created by the investment can be greater than the jobs directly employed by the new investment itself.

In turn, Keynes described this as the multiplier effect, capturing the fact investment can multiply through the economy to create an effect larger than what was initially spent.<sup>12</sup> Keynes suggested that the size of this effect would depend on the marginal propensity to consume (MPC), ie how much of their new income people would spend. The higher the MPC, the higher the multiplier. This is the case as the more money people pass on at each stage, the larger the induced effects of the investment are and the larger the total effect will be.

Michał Kalecki, an often-forgotten Polish economist, independently wrote about many of the same ideas as Keynes,<sup>13</sup> with an extra focus on how the distribution of income can affect the MPC. His concern was that an increase in business income (profits) would have a much lower MPC than household income (wages/transfers)<sup>14</sup> – therefore, how a policy affects the distribution of profits and wages can be a key determiner of how large the multiplier effects of a policy are. How different factors affect the size of multipliers is discussed in Section 2.2.

#### 2.1 HOW CAN WE MEASURE MULTIPLIERS?

When Kahn and Keynes talked about investment effects on income, the modern-day concept of gross domestic product (GDP) had yet to be invented.<sup>15</sup> Their analysis simply focused on employment and income outcomes. Kahn's analysis was solely focused on the additional employment generated from an initial public works investment or employment scheme. Keynes developed this further by looking at the effect on total income. Today, multipliers still measure the impact of additional spending on specific metrics - like consumption, household income, employment, or private investment. However, as GDP has risen in significance in economics and policymaking,<sup>16</sup> a shift to measuring the impact of spending on total market output, as GDP tries to measure, has occurred. This has been to the detriment of how household activities, mainly carried out by women, have been valued. Going forward, we will focus on multipliers that look at the response of GDP but note that alternative frameworks are needed to understand the economy holistically and properly value unpaid activities.

Fiscal multipliers are calculated as the difference in GDP caused by a change in policy divided by the cost of this policy. If the government spent £1bn on investment, which led to GDP being £2bn higher than it otherwise would be, then the multiplier would be 2. In this sense, the size of a multiplier can be seen as a measure of bang-for-buck, a multiplier of X means for every £1 you spend the economy is £X larger.

The timeframe a multiplier is measured on matters. For example, the impact of a policy in a year compared to how much was spent is often referred to as an impact multiplier. If £1bn is spent on a policy in 2024, which means 2024 GDP is £1.5bn higher than it otherwise would be, then the impact multiplier is 1.5. Alternatively, if a policy change is permanent, ie a £2bn a year increase in a public service budget, you may calculate the cumulative multiplier - that is the total amount spent over a specific period versus the total amount added to GDP each year because of it. If a policy costs £10bn over five years and after those five years the total difference in GDP each year sums to £12bn then the cumulative multiplier is 1.2. Note that the difference in GDP each year and the cost of policy can vary, allowing for different impact multipliers each year. Sometimes economists may be interested in which year the impact multiplier is at a maximum, called the peak multiplier. Throughout this report, we will specify the type of multiplier we are referring to, as this can change from study to study.

Being able to directly measure how GDP would look if a policy was or wasn't implemented is impossible. We are only able to observe what actually happens. Furthermore, we live in a complex world, such that when a policy is implemented many other factors can influence how it turns out. Therefore, trying to isolate the effect of the policy, especially on such a broad metric as GDP, is very difficult. Economists use many different methods to try to calculate multipliers, all with particular strengths and weaknesses. For example, some methods may take very detailed data to provide good estimates, whereas others may use simpler data but need to make heavy assumptions about the nature of economic relationships to get results. Some estimation techniques may automatically place restrictions on how long multiplier effects can last.<sup>17</sup> Different methods have been found to give different results for the same data and therefore

all should be considered, understanding their limitations.  $^{\rm 18,19}$ 

There is certainly more to say about how multipliers are estimated<sup>20</sup> but, importantly, acknowledging the weaknesses in multiplier estimation does not mean we should ignore estimates. Instead, it allows us to critically engage with these measures and understand their limitations. Going forward, we focus on significant and repeatable results, trying to give multiple sources for similar claims.

So far, we have spoken of multipliers in terms of policies that increase government spending and therefore have a positive impact on the rest of the economy. However, fiscal multipliers can also be measured in the opposite direction: how decreases in government spending multiply into harmful effects in the wider economy. These multipliers are still positive as the change in government spending and the change in the size of the economy go in the same direction. Such effects are important to understand how austerity/fiscal consolidation causes the economy to contract.

Also, changes in taxes alter how much money the government takes out of the economy and directly affect people's disposable incomes, so tax changes can also be considered to have multiplier effects, despite referring to spending changes thus far.

### 2.2 WHAT AFFECTS THE SIZE OF MULTIPLIERS?

Given the wide range of ways governments spend their money and the vast possibilities of potential reactions, the possible sizes of multipliers vary. In general, we might assume that multipliers will be higher the more money that is spent domestically. This is sometimes described as there being fewer 'leakages' from the domestic economy via savings, taxes, or spending on imports. Furthermore, as GDP is made up of public and private spending, multiplier effects will be larger when private spending reacts positively to public spending. Lastly, as we are measuring multiplier effects through changes in GDP, policies that focus on boosting economic output will likely have higher multipliers too.

Therefore, different characteristics can be identified that lead to spending with higher multipliers. In turn, these characteristics will help us design a framework that can give a rough idea of the expected size of the multiplier for any given policy. We now outline properties of policies that have been associated with larger multiplier effects in the literature, giving theoretical explanations of these phenomena. These characteristics are referenced in Chapter 4 where we suggest a new framework to assess multiplier effects.

### 1. Marginal propensity to consume

The marginal propensity to consume (MPC) defines how much someone will pass on when they receive extra income. Therefore, the higher the MPC, the more income someone will pass on and the higher the multiplier will be. Each time income is passed on, it may be to people with different MPCs, but so long as income is passed on enough that the sum of transactions outstrips the initial increase in income, the multiplier would be above 1. In general, poorer individuals are found to have higher MPCs.<sup>21,22</sup>

Empirical evidence shows that multipliers are higher in economies where a larger proportion of households are liquidity-constrained or live handto-mouth, spending all their income as they receive it. For example, across European economies, the average multiplier increases by 20% when a country has 25% more individuals living with liquidity constraints.<sup>23</sup> In Latin American economies, where 60% of households live paycheque-to-paycheque, the multiplier is 0.6 higher than in western economies, where only 23% of households are in such a position.<sup>24</sup>

### 2. Input-output multipliers

Kalecki speculated that the MPC out-of-profit was lower than the MPC out-of-wages.<sup>25</sup> Therefore, it is important to pay attention to what businesses may receive extra income because of a policy, as some may have higher MPCs than others leading to a higher multiplier. A way to proxy this is by looking at input-output multipliers which measure how interconnected an industry is to the rest of the economy.

Industries that rely heavily on domestic inputs tend to have higher multipliers. Spending in these industries circulates through multiple sectors, leading to greater economic impacts. In contrast, industries with higher import content see more money leak out of the domestic economy, reducing the multiplier effect. Input-output multipliers can't be used as direct fiscal multipliers as they are calculated in a different way<sup>26</sup> that ignores how the whole economy might respond to such a change, instead focusing on just the specific industry. However, they can be used to suggest spending in certain industries may lead to higher multipliers than others.<sup>27</sup>

For instance, spending in green sectors has been shown to have higher multipliers compared to more carbon-intensive sectors.<sup>28</sup> This reflects Office for National Statistics (ONS) calculations that the crude oil industry, for example, has an inputoutput multiplier of 1.6, while the electric power generation industry has an input-output multiplier of 3.1.<sup>29</sup>

### 3. Direct government expenditure (non-tax or transfer)

Government consumption and investment directly stimulate the economy in the first round but increased disposable income from tax and benefit changes do not. Government consumption and investment are directly spent on goods and services, whereas tax and benefit changes will only lead to increased economic activity once the new disposable income is spent and, in some cases, might not be spent but saved instead. Therefore, changes in taxes and benefits are likely to have more leakages from the economy in the form of savings and lags in delayed spending, whereas expenditure and investment are expected to see an immediate impact on the economy.

A meta-analysis of over 1,000 different multiplier estimates<sup>30</sup> shows that government consumption multipliers are generally close to 1, while public investment multipliers are typically higher, around 1.5. By comparison, tax cuts and benefit transfers have lower multipliers, averaging between 0.6 and 0.7.

### 4. Crowding-in or crowding-out effects

The size of the fiscal multiplier will also be influenced by its effect on private-sector spending. Other than the classic effects of how new spending is passed around the economy, new government spending may also encourage or discourage private spending. An example of why government spending could discourage private spending is investors may make decisions between investing in government bonds or other private investments, if government borrowing increases (ie more government bonds are issued) then private investment may fall as a result.<sup>31</sup> Furthermore, some suggest that households may expect increased taxes because of increased government borrowing and lower their expenditure as a result.<sup>32,33</sup> These factors are often described as public spending 'crowding out' private spending.

However, there are also reasons private spending may increase in response to increased government spending – called crowding in. Some assume crowding in happens when government expenditure lowers the cost of private consumption or investment,<sup>34,35</sup> or the public sector takes on risks that the private sector is unwilling to,<sup>36,37</sup> ie due to a new technology being unproven. This leads to higher multiplier effects as the initial spending can encourage other spending through a change in the economic environment.

An International Monetary Fund (IMF) study shows that public investment can lead to significant crowding-in effects, especially when it is targeted at infrastructure.<sup>38</sup> In such cases, multipliers can exceed 1.4 in the medium term. A European Central Bank (ECB) study shows that in countries where public debt has high amounts of foreign ownership multiplier effects are larger,<sup>39</sup> likely due to decisions to hold bonds not coming at the cost of private investment but holding other foreign debt making crowding-out effects less likely. Countries with high foreign ownership see multipliers £1–£2 higher for every £1 spent.

**5.** Changing stock of capital/size of workforce

The size of the capital stock (ie the equipment we use to create goods and services) and the number of people in the workforce also play a significant role in determining multiplier effects. Policies that increase capital (eg through investment into new equipment and technology) or the size of the workforce (eg by encouraging labour market participation) tend to have higher multipliers because they enhance the productive capacity of the economy. A larger pool of capital and labour means the economy can produce more output in response to government spending. For instance, policies that get people out of long-term unemployment can raise output significantly, especially when focused on equalising opportunities. One study in Europe finds that every extra one percentage point reduction in female long-term unemployment grows the economy by €10 per capita.<sup>40</sup> An IMF study calculates spending on green technology to have a cumulative multiplier of around 1.1–1.5 which does not decrease over five years.<sup>41</sup> Another paper finds heightened effects for green technology calculating a peak multiplier of 4.2.<sup>42</sup>

### 6. Improvements in the quality of capital or skills of the workforce

Policies that lead to improvements in the quality of capital or skills tend to generate larger multipliers over time. Policies like education, training, and health measures can improve the efficiency of the workforce while investment in new technologies can improve the efficiency of the capital stock. Together this allows for more effective use of resources and higher overall output in response to government spending. This characteristic can be differentiated from quantity changes as it should reflect the composition of the capital stock or workforce. When newer technologies replace old ones and people move into more productive jobs, this doesn't necessarily increase the size of the capital stock or workforce but does improve the quality.

For example, measures that increase education levels tend to have positive effects on growth. Education grant support to low-income students in the USA was found to have a cumulative fiscal multiplier of 2.4 after two years.<sup>43</sup> An extra year of schooling is found to increase lifetime earnings by 9%<sup>44</sup> and closing educational gaps between countries could yield permanently higher GDP by 5% for the countries catching up.<sup>45</sup> Measures that encourage technological innovation have been found to have very high multipliers. Public R&D multipliers are found to be as large as 20; a \$1 increase in spending translating to an increase in GDP of \$20 after six years. This means a longterm increase in public R&D spending of 1% could permanently increase GDP growth by 0.25%.46

### 7. General productivity enhancements

Policies that lead to greater productivity are not restricted to things that affect the quality of capital or skills of the labour force. Any policies that better enable consumption, production, and connections in the economy to take place can also improve productivity. It may be useful to make separate considerations of public capital/infrastructure in this characteristic because it can often be embedded in other production processes. For example, providing transport, energy, and communication infrastructure can make supply-chain interactions more efficient and provide businesses with more customers. Investments into infrastructure are found to have cumulative multipliers up to 1.4 by the IMF<sup>47</sup> and up to 2 in a study by University College London (UCL) academics.48

Furthermore, relieving frictions in the economy that stop people from using their skills in the best way and enabling people to follow their ambitions without barriers can also lead to improved output. For example, closing gender gaps in education and employment in Middle Eastern, North African, and South Asian countries could boost GDP growth by 2% a year.<sup>49</sup> Furthermore, social infrastructure which provides equalising opportunities by employing proportionally more women is calculated to have roughly double the impact on employment than physical infrastructure.<sup>50</sup>

### 2.3 HOW DOES ECONOMIC CONTEXT AFFECT MULTIPLIERS?

The effectiveness of fiscal multipliers varies depending on the broader economic environment. At the country level, this is well-studied.<sup>51</sup> Trade openness and exchange rate flexibility are both related to lower multipliers as they are often related to a higher prevalence of imports in the national economy, and therefore a larger source of leakages for multiplier effects.

Furthermore, the economic policies a country has in place will affect the size of multipliers too. Countries with stronger automatic stabilisers, policies that lead to increased government spending (or decreased tax collection) in economic downturns and decreased spending (or increased tax collection) in upturns, tend to have weaker multipliers.<sup>52,53</sup> Logically, if a change in government expenditure changes incomes, the automatic stabilisation should dampen the effect. For example, when incomes start to rise (because of an increase in government spending) people will start to pay more tax or lose eligibility for benefits. When incomes start to fall (because of government cuts), people will pay less tax or have their lost income compensated by social security schemes.

Lastly, multiple studies conclude that countries with higher public debt can have lower multipliers.<sup>54,55</sup> However, these studies often include developing countries where borrowing constraints can be much more real with more debt denominated in foreign currencies.<sup>56</sup> In fact, despite the USA having a high level of public debt internationally, the IMF suggests that this sort of finding does not apply,<sup>57</sup> as it is more about the safety of public debt and market reactions to extra borrowing that drive these results. In general, these studies highlight the more markets react to higher government borrowing with higher interest rates the lower multiplier effects will be, as higher interest rates will likely reduce private consumption and investment.

Therefore, while some country-level factors influence multiplier estimates, the studies we reference have mainly drawn on evidence from the UK or comparable economies. This means these factors shouldn't necessarily need to be adjusted for the above results to be interpreted in a UK context. However, we now look at some more time-specific factors that should be adjusted for, for our multiplier estimates to be relevant to the UK economy. This will help inform how we adjust multiplier estimates based on economic context in the new framework we propose in Chapter 4.

#### Position in the economic cycle

Where an economy is in the economic cycle can significantly affect the size of the multiplier. During recessions, fiscal multipliers tend to be larger because there is more unused capacity in the economy. People are more likely to experience financial constraints meaning they spend more of any additional income, especially in a financial crisis.<sup>58</sup>

According to a meta-analysis of 1,800 multiplier estimates, spending multipliers are typically 0.7 to 0.9 higher during economic downturns compared to normal periods.<sup>59</sup> Other studies show the direction of government spending may matter, with austerity measures much more harmful (and therefore exhibiting higher multipliers) in downturns than stimulus measures are effective (with multipliers not necessarily changing during recessions).<sup>60</sup> However, this result is contested by a different study suggesting stimulus can have increased multipliers in recessions, but what sort of policy is best depends on whether the downturn is driven by insufficient demand (where transfers are effective) or supply constraints (where tax cuts are more effective).<sup>61</sup>

#### **Responsiveness of monetary policy**

Another important factor influencing multipliers is the responsiveness of monetary policy. The impact of fiscal policy on the economy depends on how central banks adjust interest rates in response to changes in government spending. When central banks raise interest rates in response to fiscal expansion, they can offset the stimulus by encouraging a reduction in private consumption and investment. However, when interest rates are at or near the zero lower bound (ZLB), fiscal multipliers tend to be larger because monetary policy is less able to counteract the effects of fiscal expansion. Furthermore, explicit monetaryfiscal coordination means central banks could purposefully not respond to changes in government spending when they believe this is aligned with monetary policy objectives.

In such cases, fiscal policy has a stronger impact on GDP. One study shows under ZLB conditions, cumulative fiscal multipliers can reach up to 1.5 over four years.<sup>62</sup> Another study looking at the monetary policy position in general finds multipliers can be high as 2 when monetary policy is accommodative to the fiscal stance.<sup>63</sup>

#### 2.4 HOW LONG DO MULTIPLIERS LAST?

In general, most modern studies of fiscal multipliers focus on their short-term effects, often limited to a timeframe of less than five years. The common assumption is that fiscal policy impacts GDP in the short run but has no lasting influence on longterm output, even when changes are permanent. This stems from the belief that while government spending can boost demand in the immediate aftermath of its implementation, its effects on output diminish over time. After a period, typically five to ten years, these impacts are assumed to fade as private spending and monetary policy are assumed to respond in such a way that nullifies the impact of additional spending. The prevailing view is that long-term GDP is determined by supplyside factors, such as the amount of labour and capital employed and productivity. We now look at some other factors that can also influence how long multiplier effects last. We use these in Chapter 4 where we suggest a new framework to assess multiplier effects.

#### Supply-side factors

This idea that only supply-side factors determine GDP in the long term has a strong influence on how long the effects of policies are expected to last. For example, economists often calculate measures of potential output (ie the maximum level of GDP deemed sustainable in the long term) based solely on supply-side factors. From this, output gaps are calculated which show the difference between actual GDP and potential. However, it is important to note that there is little agreement on the size of the output gap,<sup>64</sup> how it is calculated,<sup>65</sup> when it will close,<sup>66</sup> and if the output gap and potential output are coherent measures at all.<sup>67</sup> Therefore, whether a policy is considered to have long-term effects will often depend on whether it is included in calculations of potential output. Despite this, it can still be useful to understand there are a limited number of resources in the economy and the government can only expect to have long-term effects if it stops a long-term underutilisation of resources, can permanently make resources be used more efficiently, or can increase the levels of resources available in the economy.

In this report, we purposely use the term multiplier more generally to cover any change in GDP caused by a change in government expenditure, as this can be measured over any timeline if needed. Therefore, such supply-side factors may still be useful to recognise if a policy may have longer-lasting multipliers. As referenced in Section 2.2, (5)-(7) could be seen as the characteristics of policies that affect the supply side. Therefore, in these cases, policies are often found to have permanent or long-lasting effects: a permanent 1% increase in public R&D expenditure focused on technological improvement could permanently increase GDP by 0.25%<sup>68</sup> while closing educational gaps between countries could yield a permanently higher GDP by 5% for the countries catching up.69 Public infrastructure sees stable cumulative multipliers of around 2 after six years of implementation.<sup>70</sup>

### Persistence of the policy

When a policy isn't deemed to have supply-side effects, how long it has an impact is often assumed to depend on how long the policy is implemented and when the output gap is assumed to close. A temporary change in policy, such as a one-time increase in government spending or a short-term tax cut would often be assumed to only have effects on the policy's implementation. A review of seven macroeconomic models used in western economies<sup>71</sup> suggests that this is how temporary measures are commonly modelled and this does not change for policies with higher multipliers.<sup>72</sup>

The impact of a longer-term or permanent change in policy is often assumed to fade over time - having a zero multiplier when the output gap is closed, often five to ten years after implementation.<sup>73,74</sup> This is because if a policy is not considered to have supply-side effects it will not impact the potential output. As the potential output is seen as the maximum GDP the economy can sustain, it would be assumed monetary policy and prices would adjust such that private-sector spending balances out public spending to meet the potential output by the time the output gap closes. Interestingly, when the output gap closes is often a pre-set feature of a model<sup>75</sup> and therefore isn't usually seen as dependent on fiscal policy decisions, despite recent research suggesting permanently targeting a positive output gap (GDP above potential) could have permanent positive effects.76,77

### The responsiveness of monetary policy

As explained, how monetary policy reacts is a key to when the output gap is assumed to close. When central banks are accommodative, typically by keeping interest rates low, the positive effects of fiscal expansion can persist for a longer period. Not only does the responsiveness of monetary policy affect the size of multipliers but how quickly or slowly it responds will affect how long multiplier effects last, too. In general, this can be seen as monetary policy limiting or speeding up crowdingout effects; the more responsive monetary policy is to changes in government policy the more it may affect private spending decisions. In both papers referenced in Section 2.2, the higher multipliers associated with accommodative monetary policy last as long as monetary policy stays accommodative.78,2

### Demand effects and hysteresis

Although not necessarily included in mainstream models, there may be ways to justify policies without direct supply-side effects having more long-term multiplier effects through their effect on demand. For example, even within an output gap framework, one might believe a permanent change in fiscal policy could lead to a long-lasting positive output gap<sup>80</sup> that monetary policy and private activity might not be expected to fully respond to. This at least suggests that the timing of when the output gap closes could be intertwined with fiscal policy decisions.

A useful concept to understand is hysteresis when the economic effects of a policy or a crisis can persist even after that initial policy or crisis is no longer present. For example, unemployment can have hysteresis effects, as the longer someone is unemployed the harder it may be to find a job, among other explanations.<sup>81</sup> This sort of logic can be applied to why the austerity measures implemented in Europe after the 2007-08 global financial crisis seemed to have such long-lasting effects. The unemployment and economic scarring caused by austerity measures had longlasting effects even after spending cuts were stopped or reversed. In 2012, the IMF said the impact of austerity was underestimated with low multiplier estimates to blame.<sup>82</sup> More recent research has shown that austerity's longevity was underestimated, too, with permanent multipliers around 1.2-1.583 and potential output measures being revised down because of it.84

Preventive policies, such as investments in climate change mitigation like flood defences, provide another example of how fiscal policy can generate long-term benefits. By averting future crises, these policies prevent future losses in GDP, effectively extending the life of the fiscal multiplier. For example, investments in flood defences can reduce the economic damage caused by natural disasters, leading to higher GDP in the long run. Recent analysis from the USA's Congressional Budget Office suggests for every \$1 spent on flood defences, \$2-\$6 in damages are averted.<sup>85</sup> In this context, the multiplier effect is not just about stimulating demand in the short term, but about preventing future output losses, thereby creating a lasting impact on the economy.

Lastly, demand effects may be productive. It can be hard to neatly separate the effects of supply and demand. Additional demand in the economy increases transactions which can reduce frictions in markets that slow down activity, thus increasing productivity. How long this effect lasts depends on what path the activity would have taken without additional demand. Recent research from the National Bureau of Economic Research (NBER) has shown demand increases (measured by an increase in the time people spend shopping) can lead to higher measured productivity.<sup>86</sup> Moreover, productivity, as it is usually measured, tends to be lower in economic downturns and higher in upturns.<sup>87</sup> Both these results together suggest there is a role for demand management that leads to consistently higher GDP through effects on productivity.

### 3. HOW DO GOVERNMENTS FORECAST THE IMPACT OF FISCAL POLICY?

hen governments make economic policy decisions, they provide forecasts that estimate the costs and benefits of the policy over time. This helps them understand how affordable a policy is and gives people and businesses within the economy confidence in its path. In recent times, growing numbers of governments have tried to use independent forecasters for this job,<sup>88</sup> such that forecasts aren't overly optimistic and marred by political influence. However, this has put the power of forecasting into technocratic hands. The assumptions they have made have excluded certain possibilities, which is arguably just as much of an ideological exercise but one that is protected through a veil of neutrality.<sup>89</sup> This matters, as not only do economic forecasts show the path governments will take, but they also influence the policy choices they make in the first place.

When making policy decisions, politicians will look at forecasts to decide what policies are best and where they want the economy to end up. For example, many of these independent forecasters calculate whether a country will meet its fiscal rules, ie targets of public debt and borrowing levels. If the fiscal rules are binding, policy will be adjusted until these rules can be satisfied; policies with more favourable assumptions around their impact on growth may be easier to adopt within such a self-imposed framework. Therefore, the assumptions forecasters make are an important determiner of not only what forecasts look like but also the policies that get adopted. As we focus on the effects of fiscal policy, the fiscal multiplier will be important. We now go over case studies on how different countries treat changes in fiscal policy in their models.

### **3.1 CASE STUDIES OF FISCAL FORECASTERS**

To showcase the range of assumptions employed by different forecasters, we look at five different case studies from the UK (Office for Budget Responsibility; OBR), Netherlands (Centraal Planbureau; CPB), EU (European Commission; EC), USA (Congressional Budget Office; CBO), and Canada (Office of the Parliamentary Budget Officer; PBO). Here we give an overview of the assumptions about how multipliers are applied, their size, and how long multiplier effects last.

### How are multipliers applied?

The OBR, EC, CBO, and PBO all publish tables of the multiplier assumptions they use for policies. In the UK, the OBR's multiplier assumptions only cover four different types of policies: taxes, welfare, public spending, and public investment. This means there is no differentiation between, for example, types of public spending when new policies are announced. In Canada and the EU, the multipliers are applied at a slightly more detailed level. In Canada, the PBO distinguishes housing, infrastructure, and business investment while also looking at the difference between personal and corporate tax measures. The EU provides a level of further detail with additional multipliers for consumption taxes, targeted vs general transfers, and property taxes. Lastly, the USA's CBO provides multiplier estimates bespoke to the policy being analysed. Significantly, the CBO tends to provide ranges for its multiplier estimates to reflect uncertainty over the size of its multiplier estimates.

In contrast, the CPB in the Netherlands does not calculate the direct multiplier effects of policies. Instead, it uses a macroeconomic model that relates economic variables through rules of thumb and empirical observation.<sup>90</sup> However, it does detail the response of gross domestic product (GDP) to changes in certain policies, namely value added tax (VAT), personal taxation, social transfers, and general government consumption. These can be loosely interpreted as multiplier effects assumed within the model.

All the fiscal forecasters seem to have separate capacities to calculate the economic impact of policies with supply-side impacts. At the OBR, this is done on a case-by-case basis when a policy is deemed eligible; it tends to be transparent with what it assumes in its published forecasts. However, the other fiscal forecasters are less clear on what policies are assumed to have supply-side effects and when policies are selected for bespoke modelling. Because a combination of models can be used in the final published forecast it should be noted that the multiplier effects published in methodologies may not see fruition in published forecasts.

#### What size are multiplier assumptions?

Multiplier assumptions range greatly in size across the different fiscal forecasters. This reflects the large range of results on multipliers in the empirical literature but also the lack of consensus over their value. EU and US fiscal forecasters seem to make assumptions around multipliers at the higher ends of the literature with the EC seeing its largest multiplier assumption at 1.5 and the CBO at 2.2. Interestingly, in the EC, the larger estimate is driven by a focus on the type of policy (with investment subsidies seen as the most effective and having the highest multiplier) whereas the CBO's high multiplier has to do with uncertainty over the output gap in the modelling. Therefore, even when fiscal forecasters agree on the size of multipliers their justifications will not necessarily be the same. The UK's and Canada's estimates of multipliers are more towards the middle of the literature, potentially underestimating the impact of investment. In both cases, the maximum first-year multiplier is 1 for investment measures and the minimum first-year multiplier is 0.33 in the UK and 0.1 in Canada, both reflecting the multiplier assumptions of tax policy.

Lastly, the CPB in the Netherlands has some estimates which imply negative multipliers. A negative multiplier implies an increase in government spending could lead to a fall in output. This happens under social transfers in the CPB's model. Negative multipliers are not necessarily at odds with the literature<sup>91,92</sup> but such findings often have contradictory results elsewhere in the literature<sup>93,94</sup> and tend to focus on overall government spending rather than specific policies. In the CPB's case, higher social transfers are assumed to lessen the number of people in the work force and increase unemployment numbers leading to a negative effect on GDP. Therefore, it is interesting to note not only do forecasters disagree over the size of multiplier effects but sometimes the direction of them, too.

#### How long do multipliers last?

Fiscal forecasters at the OBR and the EC both assume the output gap closes within five years. For the OBR, this means all multiplier effects diminish to zero in this time. Although the EC is less clear on how this directly effects how multipliers are applied, it probably means policies do not last longer than this. However, in both cases, supplyside policies can have longer-term effects. For example, a 2010 report from the EC suggests government investment and corporate tax policies have been modelled to have longer-term effects. As mentioned before, the OBR also does this on a case-by-case basis. Interestingly, in recent publications, the OBR has shown while it believes a 1% increase in public investment could increase growth by 2.5% over a 50-year time horizon, the impact of public investment is actually at its minimum by the end of the five-year forecast the UK bases its fiscal rules on.95

Despite the CBO having some of the largest multiplier estimates, it also thinks the effect of fiscal policy is the most short-term. CBO multiplier estimates last between one and five quarters (where ranges also cover uncertainty over how long a multiplier effect lasts). However, this may be because the CBO tends to look at temporary measures for its multiplier estimates and uses other methods for policies with long-term effects. Furthermore, the CBO is the only fiscal forecaster where the output gap is not guaranteed to close during its forecasts. This potentially allows permanent policy changes to have permanent effects, if output gaps are not deemed to close otherwise. How this adds up in final forecasts is unclear and there is a lack of transparency around assumptions here.

Interestingly, both the PBO and the CPB seem to show permanent effects of fiscal policy with Canada's PBO's assumptions increasing over time. However, this may be because the multiplier assumptions it publishes state it is assuming no monetary policy response. Therefore, how the estimates are affected by assuming a monetary response is unclear. In the Netherlands' CPB's case, policies all have permanent effects on GDP. However, it is stated that the CPB's model where such effects are taken from is only used for 1–5year forecasts, meaning such results might not be meant to be taken literally. However, it does show most policies having effects on GDP after five years which is not necessarily the case for any of the other forecasters.

Overall, again there is much variation in how fiscal forecasters approach the long-term effects of fiscal policy with some having effects that diminish after five years (or shorter) whereas others assume effects are permanent. Furthermore, there is a general lack of clarity on how supply-side measures are considered but these are likely done on a more ad hoc basis across forecasters.

### 3.2 THE IMPOSSIBLE NEUTRALITY OF FISCAL FORECASTERS

The different country examples in Section 3.1 give an idea of the diversity of approaches to multipliers and economic forecasting across independent forecasters. This shouldn't be a surprise, as choices of economic assumptions for forecasting reflect judgment calls across a wide-ranging literature. When fiscal forecasters land on multiplier assumptions, they take a stance in a way that is not readily presented or transparent. Judgments can themselves be influenced by loyalty to certain economic schools of thought which can be aligned with different political traditions. Even if this is not their intention, the multiplier assumptions forecasters choose to apply reflect an endorsement of certain types of papers and directly impact what effect fiscal policy can be expected to have.

For example, we can look at when forecasters assume multipliers below 1. For a multiplier to be below 1, the final impact on GDP must be less than what was spent by the government. For this to happen, some private spending must decrease in response to the increase in government spending. Therefore, baked into assumptions of multipliers below 1 are theories of crowding out. However, how and when crowding out happens is not overwhelmingly agreed upon<sup>96</sup> and it has certainly been an area where conflicting and contradictory results can be found.<sup>97,98</sup>

It could be said that as their multiplier assumptions are always 1 or below, the fiscal forecasters in the UK and the Netherlands (the OBR and the CPB, respectively) always assume crowding out happens. As mentioned, this takes a stance on crowding out that is not necessarily unanimous in the literature, yet little justification is provided for why the OBR or the CPB may believe crowding out is always relevant. Instead, fiscal forecasters may justify themselves by suggesting their assumptions are evidence-based and represent a neutral'middle of the literature'. Yet even if they are correct that their assumptions reflect the middle of the literature then this still endorses a finding that is controversial among the literature and is therefore non-neutral. Importantly, these assumptions pre-determine the effects of government policy in these forecasters' economic models. In this case, crowding out will make government spending look more costly, especially to politicians concerned about the debtto-GDP ratio, and this can indirectly encourage such spending to be cut.

Furthermore, the absence of hysteresis effects from most of the forecasters we have discussed is also not a neutral judgment. In some ways, this is more surprising, as hysteresis effects are more widely accepted across mainstream economics.<sup>99</sup> Yet economists tend to rely on solely supply-side models to understand long-term determinants of growth. This excludes the possibility of government policy that doesn't explicitly affect the supply side having any effect on long-term growth, meaning any debt-funded policy without supply-side effects will just increase public debt indefinitely, discouraging its adoption.

Once this idea that only supply-side policies can have long-term effects is embedded, it leads to dismissiveness of even the possibility that forecasts could be wrong for failing to account for demand deficiency. For example, in 2012, the OBR reported that its multiplier assumption would have to be 1.3 in both 2010 and 2011, roughly double what it assumed on average, to explain its forecast errors during austerity which underestimated the negative impact on GDP growth.<sup>100</sup> The idea that multipliers could be this high for this long was simply dismissed by the OBR, replaced by alternative supply-side explanations like the financial and eurozone crisis permanently altering productivity.

The problem with this is it acts as if our understanding of how economic policy affects the economy is certain, and all uncertainty comes from how the economy evolves uncontrollably. If forecast errors can only be explained through changes in economic circumstances, then there is simply no role for economic policy to improve conditions. Yet this entirely removes the possibility of how the economic context we find ourselves in and the economic policies we adopt are often interrelated. At the very least, one might expect a neutral forecaster to consider how both unexpected economic shocks and the relationship between economic policy and growth could plausibly explain errors in their forecasts. However, because the latter is ingrained as having no effect in the long term, such explanations as differing multipliers are simply ignored.

It is worth noting that the idea that only changes to the supply side of the economy can have long-term consequences came about in direct opposition to traditional Keynesian points of view.<sup>101</sup> Therefore, how supply-side policies have been considered has often excluded the use of government expenditure, instead focusing on tax cuts, deregulation, and trade liberalisation. While New Keynesian economics has tried to synthesise these points of view together, it has done so by assuming that the demand side can only have short-term effects with supply-side measures winning out in the long run. Therefore, far from being neutral, the OBR adopts a New Keynesian framework<sup>102</sup> and this framework influences how it models policies and responds to new data.

For example, while the OBR may be happy to model the long-term effects of government investment, there seems to be much less room to consider how other government expenditures could have effects on productivity. For example, the health and skills of the population or the state of education and equality can have long-term effects too, as described in Chapter 2, and this is often spearheaded by government expenditure.

Strict categories that separate government capital investment from government day-to-day spending mean higher multipliers for education and health policies won't be considered. Significantly, this can reproduce gender inequalities as it means social infrastructure (like health and care services) can often be devalued in comparison to physical infrastructure – despite some evidence finding higher multipliers in social sectors.<sup>103</sup> Therefore, not only do the underlying assumptions forecasters make limit what they see as economically possible, but the theory behind such policies can implicitly define how policies are assessed. In the OBR's case, its New Keynesian framework encourages policies to be split into their short-term demand effects and long-term supply effects. While any policy could be argued to have supply-side effects, closely following New Keynesian literature might mean only specific policies are considered to have long-term effects, the result being a framework that is actively hostile to considering some policies to have long-lasting effects.

Another reason forecasters might not want to use differing multiplier effects is they require a higher standard of evidence. Yet, this requirement for evidence can mean that innovative policies might just be modelled as having the same effects as historical averages, even when such averages might not be a good fit. While forecasters have the option to do their own ad hoc analysis or judge policies by distilled characteristics, they may be cautious and simply choose to use their default assumptions when modelling new policies. For example, such thinking might be why the OBR has failed to model the impact of net-zero investment as having supply-side effects<sup>104</sup> and instead relied on its default multipliers. Evidence of green projects having higher multipliers<sup>105</sup> and the prospects of green growth<sup>106</sup> has only come out in recent times and they may be wary of adopting new results.

However, it is worth noting that the requirement for a high standard of evidence is not necessarily equally applied. For example, the OBR has published a paper that shows how it measures the potential output and that the supply side of the economy is an inherently subjective process which requires careful judgment.<sup>107</sup> Therefore, again, the reason multiplier effects might not get such careful consideration is that the New Keynesian framework the OBR uses means such effects do not matter in the long term anyway, so little attention can be paid without much consequence for the model's results.

Obviously, not every multiplier assumption is determined in line with a given ideology. We understand forecasters have to decide on a set of multiplier assumptions. However, when multiplier assumptions are made, they should be justified and their implications explained. Forecasting bodies shouldn't just refer to what the evidence shows or doesn't show; they should explain the mechanisms and theories that produce those assumptions, with their relevance outlined. For example, if green investment is assumed to have a multiplier below 1, then it should be justified why crowding out applies in this situation. In Chapter 2, we introduced how different factors affect multipliers, while explaining the theory or logic behind such findings. Understanding not just the result but the reason a result comes about, should help forecasters and readers understand if an assumption is convincing.

### **3.3 THE INTERACTION WITH FISCAL RULES**

This all matters because fiscal forecasts serve a purpose. They are used to guide government decision-making and assess the feasibility of economic plans. The implications of different economic assumptions can alter the decisions made. In the Netherlands and Canada, fiscal forecasters provide estimates of policies during elections.<sup>108,109</sup> This likely affects both what political parties offer to the electorate and also how the electorate responds to what political parties offer. Much more directly, in the UK and the EU, fiscal rules limit debt and borrowing levels over the forecast, and therefore what assumptions are made can directly impact how much fiscal space, ie room to spend, a government has.

Therefore, when the fiscal forecasts favour certain types of assumptions they favour certain types of outcomes. For example, low and short-lasting multiplier estimates will inherently devalue public spending and mean decisions to cut government spending will be modelled to have short-lived effects. Whereas when multiplier effects are high and long-lasting, cutting spending is costly and public spending is inherently seen as more valuable.

This is important to acknowledge, as if there is any uncertainty over the actual measure of a multiplier of a particular policy then landing on a particular estimate may give it an inappropriate evaluation. This evaluation matters as it can make or break whether a policy is adopted, especially when it comes to fiscal rules. For example, say a policy costs £10bn. How much it adds to debt will depend on whether equivalent taxes are raised or whether it adds to borrowing, which will be implicit in the policy decision. However, how much it adds to GDP will depend on the forecaster's estimate of the multiplier. As fiscal rules commonly set limits around the debt-to-GDP ratio, what multiplier is assumed could even decide if the fiscal rules are met or not.

For example, recent research by academics in the EU<sup>110</sup> shows that the application of the EU's new fiscal rules rests on the assumption of a 0.75 multiplier being applied to any budget adjustments. This is especially strange given the fact the EU's other modelling allows for more varied multiplier effects and the paper it references to justify the 0.75 multiplier concludes an average multiplier of 0.8-0.9.111 The paper referenced also importantly makes a distinction between multipliers from government expenditure (between 1 and 1.4) and from tax measures (0.5 and below). In fact, applying this paper's other conclusions of multipliers up to 1.4, along with allowing for multipliers that are more persistent, the academics find that the pathways of debt-to-GDP could be significantly different than the EU currently assesses. As the EU is expected to recommend countries cut their budgets from 2025, this research shows that differing multiplier assumptions could render this exercise futile, ultimately continuing the trend where austerity has led to higher, not lower, debt-to-GDP. If spending cuts are only successful at reducing debt under certain multiplier conditions, care must be taken to acknowledge this and see what alternative measures could be taken. With higher multipliers, it's possible that declining debt could be achieved via upfront investment rather than cuts.

Furthermore, a key way multiplier assumptions can affect decision-making is how room to spend is defined under fiscal rules. In the UK, the fiscal headroom is often described as the chancellor's room to spend without breaking her fiscal rules. For example, if debt could increase from 89% to 90% of GDP before breaking the fiscal rules, the headroom would be worth 1% of GDP. However, this conception assumes that 1% of GDP is added to debt with no changes to GDP, implying a multiplier effect of zero. If instead a positive multiplier is assumed, the increased spending that adds to debt would also increase growth, allowing for higher levels of debt before reaching the 90% limit of the fiscal rules. Before the October 2024 budget, NEF analysis showed that the chancellor's spending power could increase by over £8bn by simply altering the multiplier assumption while maintaining the same level of debt.<sup>112</sup> Therefore, what multiplier assumptions are assumed can drastically change the envelope politicians allow themselves to spend and therefore careful consideration should be made if multiplier effects are being unnecessarily restrictive.

Many have suggested reforms to fiscal rules that would put a better focus on quality;<sup>113</sup> for example, making exceptions for green investment - sometimes described as a 'green golden rule'.<sup>114</sup> Under the current assumptions forecasters make about multipliers, an International Monetary Fund (IMF) study shows that increased green investment could lead to debt being forecast to grow unsustainably,<sup>115</sup> defeating the argument for a green golden rule based on debt sustainability. However, the study also suggests that differing assumptions for multipliers of green spending could change this outcome. Therefore, despite Rachel Reeves's recent reforms to the fiscal rules, the OBR's multiplier effects may still limit her ambitions, especially when these seem low compared to the potential of new policies.

### 4. THE CASE FOR A BUCKET APPROACH TO DECIDE FISCAL MULTIPLIERS

s there a better system to incorporate the use of multipliers in forecasts? An improved system should be able to encourage up-to-date thinking on multipliers, while also allowing independent forecasters to justify themselves in a way that acknowledges the implicit assumptions they make about economic policy. It should also allow them to confront uncertainty head-on, especially in a future which demands unprecedented policy changes.

### **4.1 THE IMF'S BUCKET APPROACH BY COUNTRY**

To create something akin to this system, we take inspiration from work by the International Monetary Fund (IMF). In 2014, the IMF published a paper detailing" a simple method to compute fiscal multipliers".<sup>116</sup> This method was created to address the lack of evidence of multiplier effects in many national economies. Not because these effects did not exist, but because they simply hadn't been studied or the data that would facilitate such research was limited. In fact, this paper addressed the exact problem we mentioned in Section 3.2: without an evidence-base, economists may simply ignore the effects a policy might have. It was one of the first to offer practical guidance on how to estimate multipliers when empirical estimates are not readily available.

The approach described in the IMF paper was dubbed the "bucket approach". The approach first evaluated countries by characteristics that had been found to have significant relationships with multiplier effects in the empirical literature. These characteristics included low trade openness, small automatic stabilisers, a fixed exchange rate scheme, and a low/safe level of public debt as mentioned in Section 2.3. Furthermore, the IMF suggested countries with stronger labour unions/regulations and better public expenditure management may also have higher multipliers. However, we did not reference these characteristics in Section 2.3 as the relationship between multipliers and labour union/regulations has not necessarily been held in the literature<sup>117,118</sup> and the latter criterion about public expenditure management is not an empirical finding but more a logical inference. Such results may imply that the IMF's bucket approach needs updating but as it stands these are the six criteria currently used.

Once a country has been evaluated for each characteristic, it scores one point for each time it is deemed to meet a criterion for a maximum score of six. Then, dependent on this score, a country will be assigned to a certain bucket. As the characteristics were chosen due to their relationship to multiplier effects, a high score should imply a country being associated with higher multiplier effects. Countries with higher scores are assigned to a high multiplier bucket and countries with lower scores to a low multiplier bucket.

Once countries are grouped into buckets, they can be assigned a multiplier range; for example, a country in the higher bucket may be assumed to have a default multiplier effect between 0.7 and 1.0. This range gives an approximate estimate of how changes in fiscal policy will impact gross domestic product (GDP). After the initial estimate, adjustments are made for temporary factors like the position of the economy in the economic cycle and the responsiveness of monetary policy.

The IMF's bucket approach provides a neat heuristic for estimating multiplier effects. This method can be cross-checked on countries that do have data on multipliers to show that its conclusions are sensible. The main advantage of this system is its inherent flexibility and that it allows for subjective adjustments of policies. Such flexibility is needed in uncertain economic contexts to make sure multipliers aren't rigidly based on outdated assumptions or historical averages – making clear the judgment inherent to the process.

#### **4.2 NEF'S BUCKET APPROACH BY POLICY**

Taking inspiration from the IMF's bucket approach, NEF has designed a similar approach to estimate multipliers for different policies. Instead of looking at country-level characteristics, our bucket approach looks at policy characteristics. We can also use our bucket approach to take a view on what sorts of policies will have multiplier effects that are higher but also longer lasting. One could even imagine combining the IMF's and NEF's bucket approach to get estimates of the multiplier effects of a particular policy in a particular country.

The first step in NEF's bucket approach, like the IMF's, is to judge a policy on its characteristics to get an initial multiplier category and a corresponding range of multiplier estimates. Here, we judge a policy by the characteristics laid out in Section 2.2. Table 1 could be used as a form to judge a policy, including space to justify why a policy meets that feature. In this, we generally follow the IMF's categories and ranges (ie a low multiplier bucket between 0.1 and 0.3, a mid-range multiplier bucket between 0.4 and 0.6, and a high multiplier bucket between 0.7 and 1.0). However, we make a slight adjustment to how scores correspond to the multiplier ranges, given that having just one of the characteristics in Table 1 would probably mean the multiplier is not low. Furthermore, as our characteristics have been chosen due to their strong multiplier effects, satisfying many of the criteria would likely lead to multipliers above 1 so we added a category between 1.1 and 1.4.

		Does the policy meet this criterion?		
	Is the policy	Yes/No	Justification	
1)	targeted at those with a high marginal propensity to consume?			
2)	targeted at industries with high input- output multipliers?			
3)	an expenditure measure rather than a tax/transfer?			
4)	likely to cause crowding-in effects?			
5)	likely to increase the stock of capital or the size of the workforce?			
6)	likely to improve the quality of capital or skills of the workforce?			
7)	likely to cause general productivity enhancements?			
	Total score		(Yes = 1, No = 0)	

### TABLE 1: SCORING POLICIES FOR NEF'S BUCKET APPROACH

This set of criteria allows scores in the range of 0 to 7 and could be divided into buckets as follows.

Multiplier category	Score	Multiplier range
Low multiplier	0	0.1–0.3
Mid-range multiplier	1–2	0.4–0.6
High multiplier	2–5	0.7–1.0
Very high multiplier	5–7	1.1–1.4

The overlapping of scores in different categories is intentional, as further judgment should be used to narrow in on a certain category, or uncertainty can be embraced by applying a wider multiplier range. For example, a score of 5 could translate to both the high and very high multiplier buckets, allowing for an initial estimate range between 0.7 and 1.4. Furthermore, given the characteristics of a policy can be harder to directly measure than the characteristics of an economy, one could also imagine that the total score could be a range, too. Take a government public investment programme: This might be assumed to easily meet criteria (3) and (5), but whether it would have crowdingin effects to meet criterion (4) is uncertain. This policy could be scored at least 2-3. This would mean it fits the profile of both the mid-range and high multiplier buckets and therefore the multiplier range of 0.4-1.0 could be used. Leaving this approach open to flexibility is important as it provides a transparent way to deal with uncertainty and helps encourage sensitivity analysis in economic forecasting.

To see why NEF's bucket approach is a useful tool, we can apply it to general policy instruments to see if its results agree with the literature. For example, we could consider an untargeted tax/ benefit change, a tax/benefit change targeted at the poorest, a change in government expenditure, and a change in government investment.

First, it is unlikely that an untargeted tax or benefit change would meet any of the criteria and therefore it would score 0 and be placed in the low multiplier bucket with an estimated range of 0.1–0.3. A targeted tax/benefit change would be very similar; however, it would be plausible to argue that such a policy would go to those with a higher marginal propensity to consume meeting criterion (1). It would then score 1 and be placed in the mid-range multiplier bucket, with an estimated range of 0.4–0.6.

A general increase in government expenditure would certainly meet criterion (3). Furthermore, it could be focused in an area with a high inputoutput multiplier, depending on the type of spending, or create jobs that add to the potential labour supply, satisfying criteria (2) and (5). This would score between 1 and 3 and give general government expenditure a range of 0.4–1.0. Government investment would certainly meet criterion (3) and would likely satisfy (5) as well. Furthermore, investment may target a certain industry with higher input-output multipliers or areas where private investment wants to follow the government's lead, making criteria (2) and (4) a possibility. Lastly, if the investment is designed well, it would hopefully lead to quality improvements of labour/capital or bring about productivity improvements in general satisfying criteria (6) or (7). This puts investment as likely having a score between 2 and 6, putting it in the range of midrange to very high multipliers at 0.4–1.4.

Overall, this means the bucket approach achieves a similar ordering of multipliers by fiscal instrument as the literature but the size of multipliers related to these instruments is also well reflected. Therefore, NEF's bucket approach can act as a quick way to assess the multiplier of a policy without having to rigidly categorise policies beforehand.

The next step of NEF's bucket approach is to adjust these multiplier ranges to reflect the economic context. In this instance, we do not differ from the IMF's approach in adjusting for the output gap and stance of monetary policy. The IMF suggests adjusting the multiplier up by 60% if an economy is experiencing a negative output gap (ie it is underperforming) close to its historical maximum. If the economy has a positive output gap close to its historical maximum, then the IMF suggests adjusting the multiplier down by 40%. However, we caution that how the output gap is calculated should be robustly argued. Potentially, if there is uncertainty over the size of the output gap, then a high and low adjustment could be made, with the range being extended appropriately. For example, if there is disagreement with how the output gap is measured, with some arguing the output gap is near to its maximum<sup>119</sup> (60% adjustment) and others arguing it is smaller<sup>120</sup> (20% adjustment), then the high multiplier range could be adjusted to the following: 0.84–1.6 (following a 20% adjustment on 0.7 and a 60% adjustment on 1.0).

Second, the IMF suggests adjusting the multiplier estimated upwards by 30% when monetary policy is at the zero lower bound (ZLB). If the policy is assumed to be slightly accommodative, an adjustment between 30% and 0% should be made, but this is left to the reader's judgment. Therefore, once considering monetary policy and economic cycle adjustments, the first-year multipliers from NEF's bucket approach can range from just over 0 (0.1 adjusted down by 40% for a negative output gap) to just under 3 (1.4 adjusted up by 60% for a positive output gap and by 30% for monetary policy stance). This allows for multipliers at the very top end of the empirical literature to the very bottom. Notably, both the IMF's approach and our approach exclude the possibility of negative multipliers but, as these results are rarer for individual policy assessments, it can feel appropriate to ignore the possibility for now. Furthermore, given negative multipliers imply GDP moving in an opposite direction to the change in government spending, such a result would often be thought of as direct policy failure.

The last step of NEF's bucket approach is to also consider a policy on characteristics associated with its longevity. In this case, a scoring system is not used as the impact of meeting different criteria is more complex than just increasing how long the multiplier effect lasts. Instead, we ask questions that will motivate the use of longer-term multipliers when needed as seen in Table 2.

In this system, the judgment of (a) is the most important. As discussed in Section 2.4, the persistence of a policy is likely related to how long it lasts. Temporary policies are likely to have temporary effects. For example, a policy that last two years should also affect GDP for roughly two years and, in line with research, should decrease in size over its lifetime. By default, if the answers to (b) – (d) are all no, then our suggestion for how multipliers would persist is the same as the IMF's, which assumes a temporary policy lasts as long as it is implemented and a permanent policy change lasts around five years while declining to zero. However, when the criteria for (b) – (d) are satisfied, more interesting multiplier effects may occur.

For example, if (b) is true then multiplier effects may be permanent. How, if at all, the size of the multiplier should vary over time is less clear, as some models have multiplier effects that increase over time, whereas others have ones that decrease. If a policy is deemed significant enough for (b) to be true, then forecasters should pay extra attention to it. This could help foster a more specific estimate that could overrule the bucket approach suggestion.

When (c) is true, multipliers are likely to last as long as monetary policy is accommodative. Therefore, in these scenarios, the first-year multiplier could be extended as long as monetary policy is assumed to be accommodative. It could then decline as normal when monetary policy is active again. Given the theory behind such a relationship is due to reduced crowding out at the ZLB, one would imagine this effect would only be relevant along the timeline of the policy, experiencing a multiplier effect that doesn't drop off while monetary policy is accommodative.

	Is the policy	Yes/No	Justification
a)	a permanent change?		
b)	a supply-side measure like criteria (5)–(7) in Table 1?		
C)	likely to be implemented under accommodative monetary policy?		
d)	likely to have hysteresis effects?		

### TABLE 2: QUESTIONING POLICIES BY LONG-TERM POTENTIAL FOR NEF'S BUCKET APPROACH

Lastly, if (d) is true, then even a temporary policy change could have a long-lasting multiplier effect. As discussed in Section 2.4, mainstream economists often ignore the long-term effects of demand.<sup>121</sup> Given this could have a significant effect on the size and longevity of multiplier estimates, it should be an important part of setting multiplier assumptions. Including it in our bucket approach is a way to force policymakers to engage with this idea, even if they conclude hysteresis effects are unlikely. If hysteresis effects are likely, it may be because the economy is assumed to be in a state where such effects are likely or a policy may specifically have longer-term effects on demand (eg by avoiding crisis). If a policy is deemed to have hysteresis effects, it is unclear how multipliers should be adjusted over time in this light. However, like criterion (b), if hysteresis effects are deemed significant enough for (d) to be true, then forecasters could do bespoke modelling to work out how to treat this issue, which could ultimately override the bucket approach.

### 4.3 APPLYING THE BUCKET APPROACH: THE 2024 AUTUMN STATEMENT

To see how the bucket approach could work in practice, we can apply it to the 2024 Autumn Statement. In this budget, Rachel Reeves announced an increase in spending worth 2.2% of GDP by 2029–30.<sup>122</sup> Yet, the Office for Budget Responsibility (OBR) forecasts that the economy would only be 0.15% larger by 2029–30 than if the policies hadn't been implemented at all. In fact, the OBR assumes the effects of tax measures worth 1.1% of GDP and investment crowding out drag GDP down by 0.26%. This means the total effect of government policy by 2029–30 is for the economy to be 0.11% smaller than if no policies are implemented at all.

When applying the bucket approach, instead of policies being judged by their wider categories of public spending, public investment, tax, and welfare, it would mean policies are looked at, at a more individual level. Ideally, it would mean policies are separated finely enough so that every  $\pounds$ 1 spent for a particular policy has a distinct impact. Two good examples of policies that are specific enough to assess by themselves could be the investment into the National Wealth Fund and the freeze in fuel duty which we look at criteria (1)–(7) for.

The National Wealth Fund would meet criteria (3) and (4) of the bucket approach automatically. It is a spending measure designed to crowd in private investment for the green transition. Furthermore, a good argument could be made for criterion (2), that it is targeted in areas with high input-output multipliers, as this seems to be the case for green energy in general. Similarly, we could argue the case for criteria (5) and (6) where investments will likely increase the capital stock in the economy and the quality of that capital too. This would score the policy at least five points putting it in both the high and the very high multiplier buckets – a range from 0.7 to 1.4.

Alternatively, the freeze of fuel duty would not meet criterion (3) by definition. The freeze tends to benefit richer households more<sup>123</sup> and therefore does not target people with a high marginal propensity to consume, failing to meet criterion (1). Also, the Office for National Statistics (ONS) finds the industries related to the fuel duty, for example petroleum extraction,<sup>124</sup> to have low input-output multipliers so neither would criterion (2) be met. Furthermore, it seems unlikely that the freeze in fuel duty would crowd in extra private consumption or investment more than its cost, failing to meet criterion (4). It also seems unlikely that the cut in fuel duty would have any significant effect on the quantity or quality of the capital stock or labour force or any generalised productivity improvements, failing criteria (5)–(7). Together, this might lead one to score the fuel duty cut a zero, putting it in the low multiplier bucket – a range from 0.1 to 0.3.

The application of the bucket approach already shows a slight departure from the OBR. Our estimate of the multiplier of the national wealth fund is a range (0.7–1.4) which includes the OBR's assumption in the middle (a multiplier of 1 for public investment) and our multiplier range for a fuel duty cut (0.1–0.3) would lie below the OBR's estimate (a multiplier of 0.33 for tax changes).

However, this is not the full story as our bucket approach requires adjusting these multiplier ranges for economic context. First, we would adjust for our position in the economic cycle – measured by the output gap. At the latest budget, the OBR predicted the output gap would be around 0.25% of GDP in 2024,<sup>125</sup> moving to a positive output gap (above potential output) from the policy measures within the five-year forecast before falling to zero in 2030. However, even the OBR notes that there is significant uncertainty over the size of the output gap. Therefore, for simplicity, and to aid the example of the bucket approach's application, we assume the output gap could be anywhere between its historical maximum or minimum. This implies an upward adjustment of 60% and a downward adjustment of 30%, respectively. Therefore, the range of estimates for the National Wealth Fund would now include multipliers between 0.49 and 2.24 and for fuel duty would be 0.07–0.48. Next, we would adjust for the responsiveness of monetary policy. Given interest rates are relatively high and remain so throughout the OBR's forecast we decided to not adjust here.

Lastly, policies would need to be judged for their longevity with criteria (a)–(d). In the case of the National Wealth Fund, it is likely to be permanent and it also could be judged to affect the supply side. Meeting criteria (a) and (b) should encourage careful attention to how the policy's long-term effects should be modelled. On the other hand, the fuel duty freeze is supposedly temporary (despite this being pushed back almost every budget) and we find it unlikely to cause supply-side impacts. By default, the effect of such a policy might be expected to fade.

The last two criteria, (c) and (d), could also impact how long the policies last. Criterion (c) is unlikely to be met, as interest rates are relatively high and monetary policy is likely to be responsive. Whether criterion (d) is met can be harder to discern; it might be hard to suggest there are any particular hysteresis effects pertaining to either policy. One might argue the National Wealth Fund might invest in technologies that slow down climate change, avoiding future economic damages but this argument does not seem strong enough. However, one might judge the economy to be in a state susceptible to stimulus changing its long-term impact. We leave this question open for now, but it would make sense that the National Wealth Fund be modelled with longer-lasting effects than fuel duty - both due to its expected implementation time and supply-side effects. An example of how this could be modelled is by assuming the multiplier effect of the National Wealth Fund is constant over the forecast years whereas the effect of fuel duty drops off after the freeze ends.

### FIGURE 1: FORECAST DEBT-TO-GDP MAY BE MUCH LOWER UNDER HIGHER MULTIPLIER ASSUMPTIONS

Public Sector Net Financial Liabilities (PSNFL) as a % of GDP, 2024–30, under different multiplier assumptions



Note: NEF's analysis of the OBR's October 2024 Economic and Fiscal Outlook. Green lines show multiplier assumptions from 0 to 2.3 incremented by 0.1. The red line shows the OBR's multiplier assumption.

### FIGURE 2: HIGHER MULTIPLIER EFFECTS COULD ALLOW MUCH HIGHER SPENDING WITHOUT DEBT-TO-GDP HAVING TO INCREASE





Note: NEF's analysis of the OBR's October 2024 Economic and Fiscal Outlook. We assume the multiplier is fixed over the forecast period.

Therefore, as we see from our application of the bucket approach, we may judge multipliers to be between 0.07 and 2.24 for different policies with different time scales for how long these effects last. Figure 1 shows what impact different (constant) multiplier assumptions between 0 and 2.3 would have if applied to all government spending on the new fiscal rule for net financial debt. Of course, we have already judged two policies that would fall towards the bottom and top of this range so all spending is unlikely to average at the extremities, but it shows the impact multiplier assumptions can have. For example, constant multipliers as high as 1.1 would knock 1 percentage point off the debt-to-GDP ratio by 2029–30.

As mentioned in Section 3.3, the reason this matters is our fiscal rules influence how much the chancellor may allow herself to spend. In the OBR's forecast, net financial debt as a % of GDP ends up at 83.4% of GDP. Therefore, if the multiplier effect the OBR assumed was higher, the chancellor could potentially spend more while keeping the debt-to-GDP level the same. For example, we can see how much extra spending she could have made under different multiplier assumptions without even raising debt. As shown in Figure 2, if spending was assumed to have the highest multiplier persistently then the chancellor could have spent £15.6bn a year more without the fifth-year debt-to-GDP rising at all compared to the OBR's forecast.

Overall, NEF's bucket approach suggests that a wide range of possibilities are within the realms of realistic multiplier assumptions from the OBR. And even small changes in the multiplier assumption could unlock billions to spend without debt-to-GDP rising compared to the OBR's forecast.

### **4.4 THE BENEFITS OF THE BUCKET APPROACH**

NEF's bucket approach is an easy way to assess a policy's multiplier when direct multiplier estimation is too hard or implausible. Furthermore, it provides a framework for forecasters to use to transparently justify their forecasting decisions. The bucket approach could encourage forecasters to use more up-to-date multiplier estimates, increase the transparency behind their economic assumptions while requiring them to think critically and more holistically about these assumptions, and highlight the importance of multipliers in determining policy outcomes.

As briefly mentioned in Section 4.2, the bucket approach should be able to be overridden. NEF's bucket approach is a heuristic, not an exact method, and if there is good evidence to support a particular policy having a particular multiplier then this should be preferred. However, it is a starting point for policies that do not have good evidence or where that evidence is contradictory or wideranging. The characteristics chosen for NEF's bucket approach have a significant relationship to higher multipliers in the literature. To this extent, if future literature finds evidence of new characteristics associated with higher multipliers or contradictory evidence on current characteristics, then the bucket approach could be altered. However, when adopting new characteristics, such features should try to be as distilled and mutually exclusive as possible. For example, green spending has been found to have high multipliers in the literature. However, this seems to be because green spending is more investment intensive and as a newer technology it is easier to increase the quality of the capital stock. Therefore, including green spending as a separate category would be inappropriate, as the characteristics that make it have higher multipliers are already likely covered by other criteria.

Fiscal forecasters could also adopt the bucket approach to protect themselves from political criticism in an era of growing distrust of experts.<sup>126</sup> The Liz Truss mini-budget crisis was significant, as it showed how the government might want to avoid the OBR's forecast, in case it put desired policy in a bad light. However, the reason Liz Truss wanted to avoid the OBR in the first place was that it likely wouldn't have forecasted that a budget of tax cuts would lead to enhanced economic growth, using Treasury figures for this instead.<sup>127</sup>

What was reported as a stand-off between the OBR and the Conservative party was likely the OBR not wanting to back down on significant evidence that tax cuts do very little for economic growth.<sup>128,129</sup> However, because how the OBR makes assumptions is not readily publicised, such disagreements can occur without much explanation. Instead, if the OBR adopted NEF's bucket approach, it would have a framework that would allow others to more transparently understand why it is unlikely to assume that tax cuts lead to growth.

The greatest benefit of the bucket approach is how flexible it is. It transforms economic forecasting

from something that can look exact and calculated to something that is uncertain and full of value judgments. Given the latter much more accurately describes the nature of economic forecasting, the architecture we present economic forecasts in should reflect this. The default output of NEF's bucket approach is a range of multiplier estimates, not a single value. This approach helps avoid excluding certain possibilities. In some cases, such an approach may be preferable to bespoke analysis that aims to provide certainty over new policies that are fundamentally uncertain. Furthermore, once a policy is implemented, it and its effects are more apparent. The qualitative nature of the bucket approach can help tell a story of why a policy may have been more or less successful by meeting more or fewer criteria.

The bucket approach opens new possibilities for the way some policies are assessed. Rather than assessing all day-to-day or capital spending the same way, by focusing on the characteristics of policies we can get estimates that reflect that some day-to-day spending (like that on health, education, and social care) may have higher multipliers than even some types of investment. Instead of fixed policy categories determining a policy's effects, the individual characteristics would be looked at. Ingraining such thinking in our forecasting frameworks may be useful, as an over-reliance on investment in traditional capital infrastructure may bring about GDP growth while further decoupling it from social indicators.<sup>130</sup> By explicitly allowing more flexibility in what policies can have longerterm effects we can open the door to policies that achieve growth and foster social cohesion.

Overall, the bucket approach could provide transparency in what is often a very opaque practice, and it would highlight the value judgment inherent to fiscal forecasting. Furthermore, it would allow for a diverse range of estimates to be justified without sacrificing an evidence-based approach, taking in a more holistic base of evidence by design. Most importantly, such a system could adequately respond to the fundamental uncertainty of the future by being flexible enough to estimate the impacts of new policies without having to rely on historical averages while also embracing uncertainty when needed. Together, this makes the bucket approach indispensable for assessing the policies of the future.

### 5. FURTHER RECOMMENDATIONS TO IMPROVE FORECASTING THE EFFECTS OF POLICY

urther improvements could be made to the world of fiscal forecasting so that it can better incorporate the bucket approach while also using all evidence available.

### **5.1 EMBRACING UNCERTAINTY WITH RANGES**

First, as mentioned in Chapter 4, inherent to the bucket approach is results that give ranges rather than single-point estimates. This is important for the bucket approach to give flexible results that embrace the inherent uncertainty of economic forecasting. However, the current way economic forecasts are presented does not necessarily lend itself to using assumptions that have a range of estimates associated with them. This could, however, be fixed by embracing this uncertainty and taking a more liberal approach showing multiple possible outcomes.

One way to do this would be to use fan charts. Fan charts are graphs which show how a variable may deviate from its central values over time. If forecasters wanted to implement a recommendation from the bucket approach, they could use a fan chart to show the upper and lower bounds of a variable due to the different range of assumptions used. These upper and lower bounds would be split by a central estimate which could still act as a point estimate for the forecaster if needed. However, including the upper and lower bounds allows the forecast to be interpreted in a more flexible way and means that forecasts have sensitivity analysis built into their assumptions.

Furthermore, one could imagine if presenting the upper and lower bounds of an estimate was a more common practice, then fiscal rules could reflect this too. For example, instead of fiscal rules having to be met by the exact point estimate, qualitative statements could be made about the conditions under which fiscal rules are met for a particular policy package. It could be the job of the Office for Budget Responsibility (OBR) to judge if this were realistic. For example, it could be calculated what size and how long-lasting multipliers need to be for a particular spending package to be consistent with a particular fiscal rule. This would help move the conversation from what quantity of spending meets fiscal rules to what quality of spending is needed to meet those rules. Such a change would highlight the importance of the assumptions behind policies in forecasting while changing the conversation from scrutiny over debt and borrowing limits, to scrutiny over policy design - exactly what should be the priority of fiscal policy.

### 5.2 SEEING THE FULL BENEFIT WITH LONGER HORIZONS

Another way in which economic forecasters could provide a more detailed view of economic policy is by expanding the time horizons over which they evaluate policy. For example, in its forecasts for the government budgets, the OBR is limited to a five-year forecast. It only really matters to the government if they are meeting their fiscal rules, which look just five years ahead. However, this means that policies that have benefits over a longer period simply get ignored. For example, infrastructure projects,<sup>131</sup> education,<sup>132</sup> childcare,<sup>133</sup> and welfare<sup>134</sup> policies can all have long-term effects that last well after the expenditure related to the policy has been spent. In fact, the OBR has even admitted in the past that some policies it considers have strong evidence of long-term effects, but as these do not occur in their forecast period they are simply ignored.<sup>135</sup> Furthermore, one can imagine a situation where the opposite is true, ie a policy may have positive effects in the first five years but negative effects thereafter. By not forecasting effects outside the budget timeline, important details like this are overlooked.

This wouldn't necessarily need to be done by extending the full forecast into further years. Instead, requiring forecasters to comment when there are effects that are likely to exist outside of a forecast horizon and describe this qualitatively, or through presenting different possible scenarios, could be just as informative while preventing exact estimates that give a false sense of certainty. In fact, a version of this was adopted in the October 2024 budget – with a 10-year horizon for assessing the growth impacts of policies.<sup>136</sup> In general, when and how forecasts should be extended should be flexible and based on the policies being modelled. Going forward, even a 10-year horizon may be too short, and focusing on growth effects may be too narrow.

### 5.3 A BUCKET FOR CLIMATE, INEQUALITY, AND MORE

Lastly, this report focuses on fiscal multipliers due to their controversial nature yet significant impact on economic forecasting and how fiscal rules are met. However, not all policies need to be justified by their effect on growth, as more gross domestic product (GDP) growth is not always the end goal. Indeed, when we decided to focus on fiscal multipliers that measure the response of GDP we noted how GDP's significance in economics has devalued certain economic activity, particularly women's unpaid work in the household. Our economic system requires a fundamental shakeup that includes how we measure and what we measure to track progress. Economic forecasters of the future should look at a wider range of metrics that reflect the quality of our climate, and the level of inequality in our societies, among a host of other metrics that truly reflect our livelihoods.

However, whether forecasters are looking at the impact of policy on GDP, climate, or inequality, forecasting will always be an inherently uncertain practice. Therefore, one could imagine bucket approaches being used in each of these areas. The general idea would remain the same. Policies would be assessed on their distilled characteristics, where these characteristics are chosen for their significant relationship to a desired outcome. For example, if one were to create a bucket approach for nature, one could assess policies by their potential to lower carbon emissions, reduce other pollutants, prevent natural disasters, revive natural environments, protect biodiversity, and more. This could be used to quickly assess the benefits to nature of a policy without having to do bespoke research-especially when such bespoke research would likely only provide a false sense of certainty.

Overall, there is definitely scope for economic forecasting to consider many more factors that matter for our quality of life. Yet, when economic forecasters venture into new realms, they are likely to be met with little direct evidence of the phenomenon they want to model. The bucket approach could become a powerful tool that gives forecasters no excuse to ignore the myriad effects of fiscal policy. In fact, its inherent flexibility might even make it a better tool than specific estimates which may provide too much certainty over a fundamentally uncertain phenomenon. Kahn and Keynes may have started by looking at how fiscal policy affects jobs and incomes, but today's economists should not be stuck in the past; they should use the environmental and social issues of our time to take this sort of analysis further.

### **6. CONCLUSION**

This report highlights some significant differences between the effects of fiscal policy that government forecasters assume and those from the empirical literature. These differences can dramatically affect what policies are adopted, due to how policy measures meet fiscal rules or are assessed by a debt sustainability analysis, both of which depend on multiplier assumptions.

We propose a policy-level bucket approach, inspired by the International Monetary Fund (IMF) country-level approach, to assess policies by distilled characteristics associated with higher multipliers. Policies are given points if they meet characteristics, and policies with different points are assigned to different buckets, with more points implying higher multipliers. However, an important part of this process is justifying why a policy meets a certain characteristic, with both a theoretical and empirical justification where possible.

Even if the application of the bucket approach meant a forecaster like the Office for Budget Responsibility (OBR) did not change its assumptions, it would still represent an improvement, as it would require the OBR to answer questions about why it thinks certain policies do not lead to crowding-in effects, why they are unlikely to have longer-term effects on growth, and why a classical multiplier mechanism might not apply. This would increase transparency around the rationale behind assumptions, which is often not detailed.

Yet all this begs the question of who really has the authority to decide what is right or wrong when it comes to how government spending impacts the economy? It feels that in our current system, where the fiscal forecasters make fixed assumptions about how the economy works which lead to a fixed amount politicians can spend without breaking their self-imposed fiscal rules, it is exactly the wrong way around. Instead, one could imagine a system where politicians design spending packages around the issues important to them and the fiscal forecaster assesses what effects these packages must have to meet the fiscal rules. Then the forecaster could judge if the implied effects fell within a realistic range (perhaps in the same range as what the bucket approach implies), and if not, it would give politicians a better idea of how much they could raise taxes or what spending they could prioritise.

For some time, NEF has proposed scrapping fiscal rules and replacing them with fiscal referees<sup>137</sup> who can judge the economy on more holistic measures that reflect the complexity behind debt sustainability. Such an approach could fit in nicely with the bucket approach as it would give the referees a key point of judgment - whether the assumptions underlying government plans fall within a likely range. Under such a system, the government would always be able to disagree, with the requirement to explain itself, but the referee's judgment could become a much more effective guardrail that incentivises effective policy design. Instead, if we stick to arbitrary fiscal rules, borrowing and debt limits will continue to push back progress by incentivising cuts.

Overall, fiscal forecasting needs a shakeup, to reflect on its past mistakes that ushered in austerity and to make sure it isn't a blocker to the vital investment needed today to combat climate change, the decline in public services, and below-potential growth. The evidence explored in this report provides good motivation for change. Our bucket approach would be an effective methodology to implement. We can and must forecast a better future.

### **ENDNOTES**

- 1 Van Lerven, F., Stirling, A., & Prieg, L. (2021). *Calling time: Replacing fiscal rules with fiscal referees*. New Economics Foundation. https://neweconomics.org/2021/10/calling-time
- 2 Office for Budget Responsibility. (2024). *Economic and fiscal outlook October 2024. OBR*. <u>https://obr.uk/efo/economic-and-fiscal-outlook-october-2024/</u>
- 3 Caddick, D. (2024). The OBR's fiscal powers need a rethink. New Economics Foundation. <u>https://neweconomics.org/2024/10/the-obrs-</u>fiscal-powers-need-a-rethink
- 4 Hooker, L. (2024). Budget boost to UK economy forecast to fade after two years. BBC. <u>https://www.bbc.co.uk/news/articles/</u>c5yge6knpzdo
- 5 Resolution Foundation. (2024). *More, more, more: Putting the 2024 Autumn Budget in context.* <u>https://www.resolutionfoundation.</u> org/app/uploads/2024/10/More-more-Budget-2024.pdf
- 6 Jung, C. (2024). OBR is likely underestimating the growth effects of public investment. IPPR. <u>https://www.ippr.org/articles/second-</u>round-effects
- 7 Hutton, W. (2024, November 3). Investment drives growth. That's why gloomy forecasters are so wrong about the budget. *The Observer*. https://www.theguardian.com/commentisfree/2024/nov/03/investment-drives-growth-thats-why-gloomy-forecasters-are-so-wrong-about-the-budget?CMP=Share\_iOSApp\_Other
- 8 Tims, S. (2024). Capping ambitions. New Economics Foundation. https://neweconomics.org/2024/10/capping-ambitions
- 9 Zenghelis, D. (2024). Boosting growth and productivity in the United Kingdom through investments in the sustainable economy. Grantham Research Institute. https://www.lse.ac.uk/granthaminstitute/publication/boosting-growth-and-productivity-in-theunited-kingdom-through-investments-in-the-sustainable-economy/
- 10 Trade Union Congress. (2023). From the doom loop to an economy for work not wealth. TUC. <u>https://www.tuc.org.uk/sites/default/</u>files/2023-02/Economics\_Conference\_2023\_Paper\_AW.pdf
- 11 Kahn, R. (1931). The relation of home investment to unemployment. *The Economic Journal*, 41(162). <u>https://www.jstor.org/</u> stable/2223697
- 12 Keynes, J. (1936). The general theory of employment, interest, and money. Macmillan. <u>https://link.springer.com/book/10.1007/978-3-</u> 319-70344-2
- 13 Davidson, P. (2000). There are major differences between Kalecki's theory of employment and Keynes's general theory of employment interest and money. *Journal of Post Keynesian Economics*, 23(1). https://www.jstor.org/stable/4538712
- 14 Kalecki, M. (1942). A theory of profits. The Economic Journal, 52(206/207). https://www.jstor.org/stable/2225784
- 15 Dickinsom, E. (2011, January 3). GDP: A brief history. Foreign Policy. https://foreignpolicy.com/2011/01/03/gdp-a-brief-history/
- 16 World Economic Forum. (2021). A brief history of GDP and what could come next. <u>https://www.weforum.org/agenda/2021/12/</u> stakeholder-capitalism-episode-1-a-brief-history-of-gdp/
- 17 Blanchard, O., & Quah, D. (1989). The dynamic effects of aggregate demand and supply disturbances. *The American Economic Review,*. 79(4). https://uh.edu/~bsorense/BlanchardQuah1989.pdf
- 18 Gechert, S., Paetz, C., & Villanueva, P. (2017). *Top-down vs. bottom-up? Reconciling the effects of tax and transfer shocks on output. Banco de Espana*. https://www.econstor.eu/bitstream/10419/168204/1/VfS-2017-pid-3159.pdf
- 19 Caldara, D., & Kamps, C. (2008). What are the effects of fiscal policy shocks? A VAR-based comparative analysis. European Central Bank. https://www.ecb.europa.eu/pub/pdf/scpwps/ecbwp877.pdf
- 20 Gechert, S. (2017). On theories and estimation techniques of fiscal multipliers. Forum for Macroeconomics and Macroeconomic Policies. https://www.econstor.eu/bitstream/10419/181469/1/fmm-imk-wp-11-2017.pdf
- 21 Canbary, Z., & Grant, C. (2019). The marginal propensity to consume for different socio-economic groups. Brunel University London. https://www.brunel.ac.uk/economics-and-finance/research/pdf/1916-Oct-CG-The-Marginal-Propensity-to-Consume-for-Different-Socio-economic-Groups.pdf
- 22 Fisher, J., Johnson, D., Smeeding, T., & Thompson, J. (2019). Estimating the marginal propensity to consume using the distributions of income, consumption and wealth. Federal Reserve Bank of Boston. https://www.bostonfed.org/publications/research-departmentworking-paper/2019/estimating-the-marginal-propensity-to-consume-using-the-distributions-income-consumption-wealth. aspx
- 23 Carreras, O., Kirby, S., Liadze, I., & Piggott, R. (2016). *Quantifying fiscal multipliers. National Institute for Economic and Social Research.* https://www.niesr.ac.uk/wp-content/uploads/2021/10/DP469-4.pdf
- 24 Bracco, J., Galeno, R., Juarros, L. M., Riera-Crichton, P. F., Vuletin, D., & Guillermo, J. (2021). Social transfer multipliers in developed and emerging countries. World Bank. https://documents.worldbank.org/en/publication/documents-reports/ documentdetail/327851618328958571/social-transfer-multipliers-in-developed-and-emerging-countries-the-role-of-hand-tomouth-consumers
- 25 Kalecki, M. (1942). A theory of profits. *The Economic Journal*, 52(206/207), 258–267. https://www.jstor.org/stable/2225784
- 26 Bess, R., & Ambargis, Z. (2011). Input-output models for impact analysis: Suggestions for practitioners using RIMS II multipliers. 50th Southern Regional Science Association Conference, New Orleans, LO, USA. <u>https://citeseerx.ist.psu.edu/</u> document?repid=rep1&type=pdf&doi=5af182338eb3aa00b692e4a220244f3e051a508b
- 27 Pusch, T. (2012). Fiscal spending multiplier calculations based on input-output tables an application to EU member states. *European Journal of Economics and Economic Policies*. https://www.elgaronline.com/view/journals/ejeep/9/1/article-p129.xml

- 28 Batini, N., di Serio, M, Fragetta, M., Melina, G., & Waldron, A. (2021). Building back better: How big are green spending multipliers? International Monetary Fund. https://www.imf.org/en/Publications/WP/Issues/2021/03/19/Building-Back-Better-How-Big-Are-Green-Spending-Multipliers-50264
- 29 Office for National Statistics. (2024). *UK input output analytical tables: Industry by industry. ONS.* <u>https://www.ons.gov.uk/</u>economy/nationalaccounts/supplyandusetables/datasets/ukinputoutputanalyticaltablesindustrybyindustry
- 30 Gechert, S. (2015). What fiscal policy is most effective? A meta-regression analysis. Oxford Economic Papers, 67(3). <u>https://</u> academic.oup.com/oep/article/67/3/553/2362401#83093366
- 31 Barro, R. (1974). Are government bonds net wealth? *Journal of Political Economy*, 82(6), 1095–1117. <u>https://www.jstor.org/</u> stable/1830663
- 32 Modigliani, F. (1986). Life cycle, individual thrift, and the wealth of nations. *The American Economic Review*, *76*(3), 297–313. <u>https://</u>www.jstor.org/stable/1813352
- 33 Friedman, M. (1957). A theory of the consumption function. Princeton University Press. https://www.nber.org/books-and-chapters/ theory-consumption-function
- 34 Ottonello, P., & Winberry, T. (2023). *Investment, innovation, and financial frictions. Cowles Foundation for Research in Economics.* https://cowles.yale.edu/sites/default/files/2023-06/ottonello\_winberry\_2023.pdf
- 35 Heintz, J. (2010). The impact of public capital on the US private economy: New evidence and analysis. *International Review of Applied Economics*, 24(5), 619–632. https://www.tandfonline.com/doi/abs/10.1080/02692170903426104
- 36 Azhgaliyeva, D., Beirne, J., & Mishra, R. (2022). What matters for private investment in renewable energy? *Climate Policy*, 23(1), 71–87. https://www.tandfonline.com/doi/full/10.1080/14693062.2022.2069664#abstract
- 37 Juhasz, R., Lane, N., & Rodrik, D. (2023). The new economics of industrial policy. National Bureau of Economic Research. <u>https://www.nber.org/papers/w31538</u>
- 38 Abiad, A., Furceri, D., & Topalova, P. (2015). The macroeconomic effects of public investment: Evidence from advanced economies. International Monetary Fund. <u>https://www.imf.org/en/Publications/WP/Issues/2016/12/31/The-Macroeconomic-Effects-of-PublicInvestment-Evidence-from-Advanced-Economies-42892</u>
- 39 Broner, F., Clancy, D., Erce, A., & Martin, A. (2019). *Fiscal multipliers and foreign holdings of public debt. European Central Bank.* https://www.ecb.europa.eu/pub/pdf/scpwps/ecb.wp2255~a3b6386eaa.en.pdf
- 40 Buterin, V., Fajedtic, B., & Funaric, B. (2023). Understanding the macroeconomic effects of female participation in the labour market. *Economies*, *11*(11). https://www.mdpi.com/2227-7099/11/11/280
- 41 Batini, N., di Serio, M, Fragetta, M., Melina, G., Waldron, A. (2021). Building back better: How big are green spending multipliers? International Monetary Fund. <u>https://www.imf.org/en/Publications/WP/Issues/2021/03/19/Building-Back-Better-How-Big-Are-Green-Spending-Multipliers-50264</u>
- 42 Hansa, Z. (2021). The grass is actually greener on the other side: Evidence on green multipliers from the United States. University of Cambridge. https://www.econ.cam.ac.uk/people-files/postgrad/zh274/ZeinaHasna\_JMP.pdf
- 43 De Ridder, M., Hannon, S., & Pfajfar, D. (2020). *The multiplier effect of education expenditure. Federal Reserve Board*. <u>https://www.</u>federalreserve.gov/econres/feds/files/2020058pap.pdf
- 44 Psacharopoulos, G., & Patrinos, H. (2018). Returns to investment in education: A decennial review of the global literature. *Education Economics*, 26(5). https://www.tandfonline.com/doi/abs/10.1080/09645292.2018.1484426
- 45 Hanushek, E., & Wossmann, L. (2010). Education and economic growth. *International Encyclopaedia of Education. Elsevier*. <u>https://</u>hanushek.stanford.edu/sites/default/files/publications/Hanushek%2BWoessmann%202010%20IntEncEduc%202.pdf
- 46 De Lipsis, V., Deleidi, M., Mazzucato, M., & Angolucci, P. (2023). *Macroeconomic effects of public R&D. UCL Institute for Innovation and Public Purpose*. https://www.ucl.ac.uk/bartlett/public-purpose/sites/bartlett\_public\_purpose/files/de\_lipsis\_v.\_deleidi\_m.\_ mazzucato\_m.\_and\_agnolucci\_p.\_2023\_macroeconomic\_effects\_of\_public\_rd.pdf
- 47 Abiad, A., Furceri, D., & Topalova, P. (2015). The macroeconomic effects of public investment: Evidence from advanced economies. International Monetary Fund. https://www.imf.org/en/Publications/WP/Issues/2016/12/31/The-Macroeconomic-Effects-of-PublicInvestment-Evidence-from-Advanced-Economies-42892
- 48 Deleidi, M., Lafrate, F., & Levrero, E. (2019). Public investment fiscal multipliers: an empirical assessment for European countries. UCL Institute for Innovation and Public Purpose. https://www.ucl.ac.uk/bartlett/public-purpose/sites/public-purpose/files/final\_working\_ paper\_deleidi\_iafrate\_levrero\_19\_aug.pdf
- 49 Klasen, S., & Lamanna, F. (2009). The impact of gender inequality in education and employment on economic growth: New evidence for a panel of countries. *Feminist Economics*, 15(3), 91–132. <u>https://www.tandfonline.com/doi/abs/10.1080/13545700902893106</u>
- 50 de Henau, J., & Himmelweit, S. (2020). The gendered employment gains of investing in social vs. physical infrastructure: Evidence from simulations across seven OECD countries. IKD Working Paper No. 84, The Open University. <u>https://oro.open.ac.uk/73890/1/</u> DeHenauApril2020v3.pdf
- 51 Raga, S. (2022). Fiscal multipliers: A review of fiscal stimulus options and impact on developing countries. Supporting Economic Transition. https://set.odi.org/wp-content/uploads/2022/01/Fiscal-multipliers-review.pdf
- 52 Dolls, M., Fuest, C., & Peichl, A. (2012). Automatic stabilizers and economic crisis: US vs. Europe. *Journal of Public Economics*, 96(3–4), 279–294. https://www.sciencedirect.com/science/article/abs/pii/S0047272711001642
- 53 Cohen, D., & Follette, G. (2000). The automatic stabilizers: Quietly doing their thing. Federal Reserve Bank of New York. <u>https://www.newyorkfed.org/medialibrary/media/research/epr/00v06n1/0004cohe.pdf</u>
- 54 Huidrom, R., Kose, M., Lim, J., & Ohnsorge, F. (2016). *Do fiscal multipliers depend on fiscal positions? World Bank*. <u>https://</u>documents1.worldbank.org/curated/en/928971467137028237/pdf/WPS7724.pdf

- 55 Ilzetzki, E., Mendoza, E., & Végh, C. (2013). How big (small?) are fiscal multipliers? *Journal of Monetary Economics*, 60(2), 239–254. https://www.sciencedirect.com/science/article/abs/pii/S030439321200116X
- 56 Sims, C. (1999). Domestic currency denominated government debt as equity in the primary surplus. Latin American meetings of the Econometric Society. http://sims.princeton.edu/yftp/Cancun/DebtEquity.pdf
- 57 Batini, N., Eyraud, L., & Weber, A. (2014). *A simple method to compute fiscal multipliers. International Monetary Fund.* <u>https://www.</u> imf.org/en/Publications/WP/Issues/2016/12/31/A-Simple-Method-to-Compute-Fiscal-Multipliers-41627
- 58 Corsetti, G., Meier, A., & Muller, G. (2012). What determines government spending multipliers? International Monetary Fund. <u>https://</u> www.imf.org/external/pubs/ft/wp/2012/wp12150.pdf
- 59 Gechert, S., & Rannenberg, A. (2018). Which fiscal multipliers are regime-dependent? A meta-regression analysis. *Journal of Economic Surveys*, 32(4), 1160–1182. https://onlinelibrary.wiley.com/doi/abs/10.1111/joes.12241
- 60 Barnichon, R., Debortoli, D., & Matthes, C. (2022). Understanding the size of the government spending multipliers: It's in the sign. *The Review of Economic Studies, 89(1), 87–117*. https://academic.oup.com/restud/article-abstract/89/1/87/6294943
- 61 Ghassibe, M., & Zanetti, F. (2022). State dependence of fiscal multipliers: The source of fluctuation matters. *Journal of Monetary Economics*, 123, 1–23. https://www.sciencedirect.com/science/article/abs/pii/S0304393222001155
- 62 Klein, M., & Winkler, R. (2018). The government spending multiplier at the zero lower bound: International evidence from historical data. University of Antwerp. https://ideas.repec.org/p/ant/wpaper/2018001.html
- 63 Cloyne, J., Jordà, O., & Taylor, A. (2021). *Decomposing the fiscal multiplier*. *National Bureau for Economic Research*. <u>https://www.nber.</u> org/papers/w26939
- 64 Chen, J., & Górnicka, L. (2020). Measuring output gap: is it worth your time? International Monetary Fund. <u>https://papers.ssrn.com/</u> sol3/papers.cfm?abstract\_id=3553185
- 65 Schuster, F., Krahé, M., Sigl-Glöckner, P., & Leusder, D. (2021). *The cyclical component of the debt brake: Analysis and a reform* proposal. Dezernat Zukunft. https://www.dezernatzukunft.org/wp-content/uploads/2021/11/Konjunkturkomponente\_eng\_final-2. pdf
- 66 Office for Budget Responsibility. (2013). Is it plausible to assume a negative output gap after five years? OBR. <u>https://obr.uk/box/is-it-</u>plausible-to-assume-a-negative-output-gap-after-five-years/
- 67 Tily, G. (2019). From false multipliers to 'nonsense output gaps'. Progressive Economy Forum. <u>https://progressiveeconomyforum.com/</u> wp-content/uploads/2019/10/Tily-Multiplier-and-nonsense-output-gaps.pdf
- 68 De Lipsis, V., Deleidi, M., Mazzucato, M., & Angolucci, P. (2023). *Macroeconomic effects of public R&D. UCL Institute for Innovation and Public Purpose*. https://www.ucl.ac.uk/bartlett/public-purpose/sites/bartlett\_public\_purpose/files/de\_lipsis\_v.\_deleidi\_m.\_ mazzucato\_m.\_and\_agnolucci\_p.\_2023\_macroeconomic\_effects\_of\_public\_rd.pdf
- 69 Hanushek, E., & Wossmann, L. (2010). Education and economic growth. *International Encyclopaedia of Education. Elsevier*. <u>https://</u>hanushek.stanford.edu/sites/default/files/publications/Hanushek%2BWoessmann%202010%20IntEncEduc%202.pdf
- 70 Deleidi, M., Lafrate, F., & Levrero, E. (2019). Public investment fiscal multipliers: An empirical assessment for European countries. UCL Institute for Innovation and Public Purpose. https://www.ucl.ac.uk/bartlett/public-purpose/sites/public-purpose/files/final\_working\_ paper\_deleidi\_iafrate\_levrero\_19\_aug.pdf
- 71 Coenen, G., Erceg, C., Freedman, C., Furceri, D., Kumhof, M., Lalonde, R., Laxton, D., ...Veld, J. (2010). *Effects of fiscal stimulus in structural models. International Monetary Fund.* https://www.imf.org/external/pubs/ft/wp/2010/wp1073.pdf
- 72 Anderson, D., Hunt, B. L., Kortelainen, M., Kumhof, M., Laxton, D., Muir, D.V., Mursula, S., & Snudden, S. (2013). *Getting to know GIMF: The simulation properties of the global integrated monetary and fiscal model. International Monetary Fund.* <u>https://www.</u> imf.org/en/Publications/WP/Issues/2016/12/31/Getting-to-Know-GIMF-The-Simulation-Properties-of-the-Global-Integrated-Monetary-and-Fiscal-40357
- 73 European Commission. (2010). Public finances in EMU 2010. <u>https://ec.europa.eu/economy\_finance/publications/european\_</u>economy/2010/pdf/ee-2010-4\_en.pdf
- 74 Anderson, D., Hunt, B. L., Kortelainen, M., Kumhof, M., Laxton, D., Muir, D.V., Mursula, S., & Snudden, S. (2013). Getting to know GIMF: The simulation properties of the global integrated monetary and fiscal model. International Monetary Fund. <u>https://www.imf.org/en/Publications/WP/Issues/2016/12/31/Getting-to-Know-GIMF-The-Simulation-Properties-of-the-Global-Integrated-Monetary-and-Fiscal-40357</u>
- 75 Office for Budget Responsibility. (2013). *Is it plausible to assume a negative output gap after five years? OBR*. <u>https://obr.uk/box/is-it-</u>plausible-to-assume-a-negative-output-gap-after-five-years/
- 76 Furlanetto, F., Lepetit, A., Robstad, Ø., Rubio-Ramírez, J., & Uldeval, P. (2021). *Estimating hysteresis effects. Federal Reserve Board.* https://www.federalreserve.gov/econres/feds/files/2021059pap.pdf
- 77 Cerra, V., Fatas. A., & Saxena, S. (2020). *Hysteresis and business cycles. International Monetary Fund.* <u>https://www.imf.org/en/</u> Publications/WP/Issues/2020/05/29/Hysteresis-and-Business-Cycles-49265
- 78 Cloyne, J., Jordà, O., & Taylor, A. (2021). Decomposing the fiscal multiplier. National Bureau for Economic Research. <u>https://www.nber.org/papers/w26939</u>
- 79 Klein, M., & Winkler, R. (2018). The government spending multiplier at the zero lower bound: International evidence from historical data. University of Antwerp. https://ideas.repec.org/p/ant/wpaper/2018001.html
- 80 Cerra, V., Fatas. A., & Saxena, S. (2020). *Hysteresis and business cycles. International Monetary Fund.* <u>https://www.imf.org/en/</u> Publications/WP/Issues/2020/05/29/Hysteresis-and-Business-Cycles-49265
- 81 Blanchard, O., & Summers, L. (1986). Hysteresis in unemployment. National Bureau of Economic Research. <u>https://www.nber.org/</u> papers/w2035

- 82 Blanchard, O., & Leigh, D. (2013). Growth forecast errors and fiscal multipliers. International Monetary Fund. <u>https://www.imf.org/</u> external/pubs/ft/wp/2013/wp1301.pdf
- 83 Gechert, S., Horn, G., & Paetz, C. (2017). Long-term effects of fiscal stimulus and austerity in Europe. Macroeconomic Policy Institute. https://www.boeckler.de/en/faust-detail.htm?sync\_id=7815
- 84 Fatas, A., & Summers, L. (2016). The permanent effects of fiscal consolidations. National Bureau of Economic Research. <u>https://www.nber.org/system/files/working\_papers/w22374/w22374.pdf</u>
- 85 Congressional Budget Office. (2024). Federal spending for flood adaptions. https://www.cbo.gov/publication/59971
- 86 Bai, Y., Rios-Rull, J-V., & Storesletten, K. (2024). *Demand shocks as technology shocks*. *National Bureau of Economic Research*. <u>https://</u>www.nber.org/system/files/working\_papers/w32169/w32169.pdf
- 87 Field, A. (2007). Procyclical TFP and the cyclicality of growth in output per hour, 1890-2004. Santa Clara University. <u>https://www.</u>mcgill.ca/economics/fieldpaper.pdf
- 88 Von Trapp, L., Lienert, I., & Wehner, J. (2016). Principles for independent fiscal institutions and case studies. OECD. <u>https://personal.lse.ac.uk/wehner/ifis.pdf</u>
- 89 Clift, B. (2023). The politics of technocratic fiscal oversight in Britain. *London School of Economics Blog.* <u>https://blogs.lse.ac.uk/</u> politicsandpolicy/the-politics-of-technocratic-fiscal-oversight-in-britain/
- 90 CPB Economic Policy Analysis. (2021). Saffier 3.0: Technical background. https://www.cpb.nl/en/saffier-3-0-technical-background
- 91 Alesina, A., & Ardagna, S. (2009). Large changes in fiscal policy: Taxes versus spending. National Bureau of Economic Research. <u>https://</u> www.nber.org/system/files/working\_papers/w15438/w15438.pdf
- 92 Nie, O. (2020). Expansionary fiscal austerity. World Bank. https://documents1.worldbank.org/curated/en/430691596113581519/pdf/ Expansionary-Fiscal-Austerity-New-International-Evidence.pdf
- 93 Breuer, C. (2019). Expansionary austerity and reverse causality: A critique of the conventional approach. Institute for New Economic Thinking. https://www.ineteconomics.org/uploads/papers/WP\_98-Breuer.pdf
- 94 Botta, A. (2016). The theoretical weaknesses of the expansionary austerity doctrine. University of Greenwich. <u>https://gala.gre.ac.uk/id/</u>eprint/14446/1/GPERC34\_Botta\_V2.pdf
- 95 Suresh, N., Ghaw, R., Obeng-Osei, R., Wickstead, T. (2024). Public investment and potential output. The Office for Budget Responsibility. https://obr.uk/docs/dlm\_uploads/Public-investment-and-potential-output\_August-2024.pdf
- 96 Friedman, B. (1978). Crowding out or crowding in? The economic consequences of financing government deficits. Brookings Papers on Economic Activity. https://www.brookings.edu/wp-content/uploads/2016/11/1978c\_bpea\_friedman.pdf
- 97 Cevik, S. (2019). You are suffocating me! Firm-level evidence on crowding-out. International Monetary Fund. <u>https://www.elibrary.imf.</u> org/view/journals/001/2019/080/article-A001-en.xml
- 98 Abiad, A., Furceri, D., & Topalova, P. (2015). The macroeconomic effects of public investment: Evidence from advanced economies. International Monetary Fund. https://www.imf.org/en/Publications/WP/Issues/2016/12/31/The-Macroeconomic-Effects-of-PublicInvestment-Evidence-from-Advanced-Economies-42892
- 99 Cerra, V., Fatas. A., & Saxena, S. (2020). *Hysteresis and business cycles. International Monetary Fund.* <u>https://www.imf.org/en/</u> Publications/WP/Issues/2020/05/29/Hysteresis-and-Business-Cycles-49265
- 100 Office for Budget Responsibility. (2012). Forecast evaluation report. OBR. <u>https://assets.publishing.service.gov.uk/</u> media/5a755ad3ed915d6faf2b2587/9780108511981.pdf
- 101 Feldstein, M. (1986). Supply side economics: Old truths and new claims. National Bureau of Economic Research. <u>https://www.nber.org/</u> system/files/working\_papers/w1792.pdf
- 102 Clift, B. (2023). The OBR, fiscal forecasting, and the politics of economic method. Oxford University Press. <u>https://academic.oup.com/</u> book/45821/chapter-abstract/400925065?redirectedFrom=fulltext
- 103 de Henau, J., & Himmelweit, S. (2020). The gendered employment gains of investing in social vs. physical infrastructure: Evidence from simulations across seven OECD countries. IKD Working Paper No. 84, The Open University. <u>https://oro.open.ac.uk/73890/1/</u> DeHenauApril2020v3.pdf
- 104 Office for Budget Responsibility. (2021). *Climate-related measures in the Budget and Spending Review*. <u>https://obr.uk/box/climate-</u>related-measures-in-the-budget-and-spending-review/
- 105 Batini, N., di Serio, M, Fragetta, M., Melina, G., Waldron, A. (2021). Building back better: How big are green spending multipliers? International Monetary Fund. <u>https://www.imf.org/en/Publications/WP/Issues/2021/03/19/Building-Back-Better-How-Big-Are-Green-Spending-Multipliers-50264</u>
- 106 Valero, A., & Zenghelis, D. (2024). Sustainable investment is good macroeconomics and fiscally prudent for the UK. Grantham Research Institute. https://www.lse.ac.uk/granthaminstitute/news/sustainable-investment-is-good-macroeconomics-and-fiscally-prudentfor-the-uk/
- 107 Murray, J. (2014). Output gap measurement: Judgement and uncertainty. Office for Budget Responsibility. <u>https://obr.uk/docs/dlm\_</u>uploads/WorkingPaperNo5.pdf
- 108 CPB Economic Policy Analysis. (2023). Elections 2023. https://www.cpb.nl/en/elections-2023
- 109 Office of the Parliamentary Budget Officer. (2021). *Election proposal costing*. <u>https://www.pbo-dpb.ca/en/epc-estimates-</u>estimations-cpe
- 110 Heimberger, P., Welslau, L., Schütz, B., Gechert, S., Guarascio, D., & Zezza, F. (2024). Debt sustainability analysis in reformed EU fiscal rules: The effect of fiscal consolidation on growth and public debt ratios. *Intereconomics*, *59*(*5*), *276–283*. <u>https://www.</u> intereconomics.eu/contents/year/2024/number/5/article/debt-sustainability-analysis-in-reformed-eu-fiscal-rules.html

- 111 Carnot, N., & de Castro, F. (2015). The discretionary fiscal effort: An assessment of fiscal policy and its output effect. European Commission. https://ec.europa.eu/economy\_finance/publications/economic\_paper/2015/pdf/ecp543\_en.pdf
- 112 Caddick, D. (2024). The OBR's fiscal powers need a rethink. New Economics Foundation. https://neweconomics.org/2024/10/the-obrsfiscal-powers-need-a-rethink
- 113 Fiscal Matters. (2022). We need a quantum leap in EU economic fiscal governance for a future-fit economy. <u>https://eeb.org/open-letter-fiscal-reform/</u>
- 114 Darvas, Z. (2022). A European climate fund or a green golden rule: Not as different as they seem. Bruegel. https://www.bruegel.org/ blog-post/european-climate-fund-or-green-golden-rule-not-different-they-seem
- 115 Caselli, F., Lagerborg, A., & Medas, P. (2024). Green fiscal rules? Challenges and policy alternatives. International Monetary Fund. https://www.elibrary.imf.org/view/journals/001/2024/125/001.2024.issue-125-en.xml
- 116 Batini, N., Eyraud, L., & Weber, A. (2014). A simple method to compute fiscal multipliers. International Monetary Fund. <u>https://www.</u> imf.org/en/Publications/WP/Issues/2016/12/31/A-Simple-Method-to-Compute-Fiscal-Multipliers-41627
- 117 Boeck, M., Crespo Cuaresma, J., & Glocker, C. (2022). Labor market institutions, fiscal multipliers, and macroeconomic volatility. Centre for Economic Studies and ifo Institute. https://www.econstor.eu/bitstream/10419/260879/1/cesifo1\_wp9749.pdf
- 118 Cacciatore, M., Duval, R., Furceri, D., & Zdzienicka, A. (2021). Fiscal multipliers and job-protection regulation. *European Economic Review*, 132, 103616. https://www.sciencedirect.com/science/article/abs/pii/S0014292120302464?via%3Dihub
- 119 Kaminska, I., & Jones, C. (2020, May 4). An absolutely enormous output gap is forming. *Financial Times*. <u>https://www.ft.com/</u> content/27ee80b4-73d2-4daa-a378-99f7bec2ef64
- 120 Office for Budget Responsibility. (2024). Potential output and the output gap. https://obr.uk/forecasts-in-depth/the-economyforecast/potential-output-and-the-output-gap/#potential
- 121 Hein, E. (2023). Varieties of demand and growth regimes post-Keynesian foundations. *European Journal of Economics and Economic Policies*. https://www.elgaronline.com/view/journals/ejeep/20/3/article-p410.xml
- 122 Office for Budget Responsibility. (2024). *Economic and fiscal outlook October 2024. OBR.* <u>https://obr.uk/efo/economic-and-fiscal-outlook-october-2024/</u>
- 123 Chapman, A. (2023). Carbon emissions from fuel duty freeze equivalent to putting extra 2M cars on the road. New Economics Foundation. https://neweconomics.org/2023/03/carbon-emissions-from-fuel-duty-freeze-equivalent-to-putting-extra-2m-cars-on-the-road
- 124 Office for National Statistics. (2024). *UK input-output analytical tables: industry by industry*. <u>https://www.ons.gov.uk/economy/</u>nationalaccounts/supplyandusetables/datasets/ukinputoutputanalyticaltablesindustrybyindustry
- 125 Office for Budget Responsibility. (2024). *Economic and fiscal outlook October 2024*. <u>https://obr.uk/efo/economic-and-fiscal-outlook-october-2024/</u>
- 126 Portes, R. (2017). "I think the people of this country have had enough of experts". Think at London Business School. <u>https://www.london.</u> edu/think/who-needs-experts
- 127 HM Treasury. (2022). The Growth Plan 2022. https://www.gov.uk/government/publications/the-growth-plan-2022-documents/ the-growth-plan-2022-html#policy-decisions
- 128 Gechert, S., & Heimberger, P. (2022). Do corporate tax cuts boost economic growth? *European Economic Review*, 147, 104157. https://www.sciencedirect.com/science/article/pii/S0014292122000885
- 129 Gale, W., & Samwick, A. (2016). Effects of income tax changes on economic growth. Brookings Institute. <u>https://www.brookings.edu/</u> articles/effects-of-income-tax-changes-on-economic-growth/
- 130 Office for National Statistics. (2024). *Measuring progress, well-being and beyond GDP in the UK: May 2024*. https://www.ons.gov.uk/ peoplepopulationandcommunity/wellbeing/bulletins/measuringprogresswellbeingandbeyondgdpintheuk/may2024
- 131 Bivens, J. (2017). The potential macroeconomic benefits from increasing infrastructure investment. Economic Policy Institute. <u>https://www.</u>epi.org/publication/the-potential-macroeconomic-benefits-from-increasing-infrastructure-investment/
- 132 Psacharopoulos, G., Patrinos, H. (2018). Returns to investment in education: A decennial review of the global literature. *Education Economics*, 26(5), 445–458. https://www.tandfonline.com/doi/abs/10.1080/09645292.2018.1484426
- 133 Sandher, J., & Stephens, T. (2023). Investing in universal early years education pays for itself. New Economics Foundation. <u>https://</u>neweconomics.org/2023/07/investing-in-universal-early-years-education-pays-for-itself
- 134 Tims, S. (2024). Capping ambitions. New Economics Foundation. https://neweconomics.org/2024/10/capping-ambitions
- 135 Office for Budget Responsibility. (2019). Forecast evaluation report. <u>https://obr.uk/docs/dlm\_uploads/Forecast\_evaluation\_report\_</u> December\_2019-1.pdf
- 136 Reeves, R. (2024). Autumn Budget 2024 speech. HM Treasury. https://www.gov.uk/government/speeches/autumn-budget-2024speech
- 137 Van Lerven, F., Stirling, A., & Prieg, L. (2021). *Calling time: Replacing fiscal rules with fiscal referees. New Economics Foundation.* https://neweconomics.org/2021/10/calling-time

#### WWW.NEWECONOMICS.ORG

info@neweconomics.org +44 (0)20 7820 6300 @NEF Registered charity number 1055254

### COVER IMAGE BY:

Alphotographic (iStock Photos)

### **PUBLISHED:**

January 2025

NEF is a charitable think tank. We are wholly independent of political parties and committed to being transparent about how we are funded.

### WRITTEN BY:

Dominic Caddick, Chaitanya Kumar

### **SPECIAL THANKS TO**

We are grateful for the external experts and internal NEF team who helped greatly in making this report. NEF is also thankful to the support of ClimateWorks Foundation which made this work possible.

